RESEARCH CONTRIBUTION



Cohomology Rings of Toric Bundles and the Ring of Conditions

Johannes Hofscheier¹ · Askold Khovanskii^{2,3} · Leonid Monin⁴

Received: 23 November 2022 / Revised: 30 June 2023 / Accepted: 1 July 2023 / Published online: 14 August 2023 © The Author(s) 2023

Abstract

The celebrated BKK Theorem expresses the number of roots of a system of generic Laurent polynomials in terms of the mixed volume of the corresponding system of Newton polytopes. In Pukhlikov and Khovanskii (Algebra i Analiz 4(4):188-216, 1992), Pukhlikov and the second author noticed that the cohomology ring of smooth projective toric varieties over $\mathbb C$ can be computed via the BKK Theorem. This complemented the known descriptions of the cohomology ring of toric varieties, like the one in terms of Stanley-Reisner algebras. In Sankaran and Uma (Comment Math Helv 78(3):540–554, 2003), Sankaran and Uma generalized the "Stanley-Reisner description" to the case of toric bundles, i.e., equivariant compactifications of (not necessarily algebraic) torus principal bundles. We provide a description of the cohomology ring of toric bundles which is based on a generalization of the BKK Theorem, and thus extends the approach by Pukhlikov and the second author. Indeed, for every cohomology class of the base of the toric bundle, we obtain a BKK-type theorem. Furthermore, our proof relies on a description of graded-commutative algebras which satisfy Poincaré duality. From this computation of the cohomology ring of toric bundles, we obtain a description of the ring of conditions of horospherical homogeneous spaces as well as a version of Brion-Kazarnovskii theorem for them. We conclude the manuscript with a

To the memory of Ernest Borisovich Vinberg

☑ Leonid Monin leonid.monin@epfl.ch

> Johannes Hofscheier johannes.hofscheier@nottingham.ac.uk

Askold Khovanskii askold@math.utoronto.ca

- ¹ School of Mathematical Sciences, University of Nottingham, Nottingham NG7 2RD, UK
- ² Department of Mathematics, University of Toronto, Toronto, Canada
- ³ Moscow Independent University, Moscow, Russia
- ⁴ Institut de Mathématiques, EPFL, Bâtiment MA, Station 8, 1015 Lausanne, Switzerland

number of examples. In particular, we apply our results to toric bundles over a full flag variety G/B. The description that we get generalizes the corresponding description of the cohomology ring of toric varieties as well as the one of full flag varieties G/B previously obtained by Kaveh (J Lie Theory 21(2):263–283, 2011).

Keywords Toric varieties · Toric bundles · Spherical varieties · Newton polyhedra · Newton–Okounkov bodies · Singular cohomology · Ring of conditions

Mathematics Subject Classification Primary 14M25 · 52B20; Secondary 14C17 · 14N10

1 Introduction

In this paper, we discuss descriptions of the cohomology ring of toric bundles and how to use them to compute the ring of conditions of horospherical homogeneous spaces. Our main contributions are:

- (1) A generalization of the celebrated BKK Theorem (Theorem 4.1). We also provide an alternative interpretation (Theorem 4.3). In particular, this yields a convex theoretic formula for the self-intersection polynomial on the second cohomology of a toric bundle (Theorem 9.2).
- (2) A description of (not necessarily finite-dimensional) graded commutative algebras with Poincaré duality (Theorem 7.16).
- (3) A description of the cohomology ring of toric bundles (Theorem 8.1). In particular, we retrieve the early description by Uma and Sankaran (see the proof of Theorem 3.8 given in Sect. 8.1).
- (4) A description of the ring of conditions of horospherical homogeneous spaces (Corollary 8.4).
- (5) We illustrate our results by reproving the Brion–Kazarnovskii theorem for horospherical varieties (Theorem 10.4). We complement the illustration with the computation of the cohomology ring of some toric bundles over full flag varieties and the corresponding ring of conditions (Theorem 10.13).

Let us start with a recollection of the classical toric case. Let $T \simeq (\mathbb{C}^*)^n$ be an algebraic torus with character lattice M and lattice of one-parameter subgroups N. Further, let X_{Σ} be a smooth projective T-toric variety given by a fan $\Sigma \subseteq N_{\mathbb{R}} := N \otimes_{\mathbb{Z}} \mathbb{R}$. Denote the rays of Σ by $\Sigma(1) = \{\rho_1, \ldots, \rho_s\}$ and their primitive generators in N by e_1, \ldots, e_s . Recall the following well-known description of the cohomology ring of X_{Σ} (see, for instance, [11, Theorem 12.4.1]):

$$H^*(X_{\Sigma},\mathbb{R})\simeq \mathbb{R}[x_1,\ldots,x_s]/(I+J)=:R_{\Sigma},$$

where *I* is generated by monomials $x_{i_1} \cdots x_{i_t}$, such that $\rho_{i_1}, \ldots, \rho_{i_t} \in \Sigma(1)$ are distinct and do not form a cone in Σ and $J = \langle \sum_{i=1}^{s} \chi(e_i) x_i : \chi \in M \rangle$. Note that *I* depicts the Stanley–Reisner ideal of the fan Σ and, therefore, we refer to this description as the *Stanley–Reisner description* of $H^*(X_{\Sigma}, \mathbb{R})$. This description yields an algorithm to evaluate products of cohomology classes in the top degree of R_{Σ} . In Sect. 2.1, we provide further details.

On the other hand, given line bundles L_1, \ldots, L_t on X_{Σ} , one can directly compute a top degree intersection product $c_1(L_1)^{k_1} \cdots c_1(L_t)^{k_t}$ in $H^*(X_{\Sigma}, \mathbb{R})$ using the BKK theorem [29] (see also [3, 5]). Here, $c_1(L_i) \in H^2(X_{\Sigma}, \mathbb{R})$ denotes the first Chern class of L_i . More precisely, as any line bundle on X_{Σ} is the difference of two ample line bundles, it suffices to evaluate products $c_1(L_1)^{k_1} \cdots c_1(L_t)^{k_t}$ with all L_i ample. By the toric dictionary, the ample line bundles L_i correspond to polytopes P_i whose normal fan coarsens the fan Σ . Using the BKK Theorem, we obtain

$$c_1(L_1)^{k_1}\cdots c_1(L_t)^{k_t} = n! \cdot V(\underbrace{P_1,\ldots,P_1}_{k_1 \text{ times}},\ldots,\underbrace{P_t,\ldots,P_t}_{k_t \text{ times}}),$$

where $V(P_1, \ldots, P_1, \ldots, P_t, \ldots, P_t)$ denotes the mixed volume of the *n*-tuple $(P_1, \ldots, P_1, \ldots, P_t, \ldots, P_t)$. In Ref. [33, Section 1.4], Pukhlikov and the second author observed that the information on these intersection products suffices to regain a description of the cohomology ring $H^*(X_{\Sigma}, \mathbb{R})$. As virtual polytopes play a crucial role in this description, we refer to it as the *virtual polytope description* of the cohomology ring $H^*(X_{\Sigma}, \mathbb{R})$. In Sect. 2.2, we provide further details.

In Ref. [21], Kaveh used a similar approach to study the cohomology ring of spherical varieties provided that it is generated in degree 2. In particular, he obtained a "volume-polynomial-description" for the cohomology ring of full flag varieties G/Busing the relation between the volume of string polytopes and the degree of the corresponding line bundle from [1]. Recently, in Ref. [24], Kaveh and the second author introduced the *ring of complete intersections* for an (arbitrary) algebraic variety. For a smooth complete algebraic variety X whose cohomology ring $H^*(X, \mathbb{R})$ is generated in degree 2, the ring of complete intersections coincides with $H^*(X, \mathbb{R})$. They also provide a description of the ring of complete intersections which generalizes the results of [21].

Certainly, the cohomology ring of toric varieties accepts many more descriptions. In particular, we want to emphasize the description by Brion [8]. Brion's description is a byproduct of the computation of the *equivariant* cohomology ring. In Ref. [8], Brion proves that $H^*(X_{\Sigma}, \mathbb{R})$ is isomorphic to the ring of piecewise polynomial functions on Σ modulo the ideal generated by global linear functions.

Our main contribution is a generalization of this picture to the case of toric bundles. For a principal *T*-torus bundle $p: E \rightarrow B$ over a closed orientable real manifold *B* and a *T*-toric variety X_{Σ} , let $E_{\Sigma}:=(E \times X_{\Sigma})/T$ be the associated toric bundle (see Sect. 3 for details). A generalization of the Stanley–Reisner description for the cohomology ring $H^*(E_{\Sigma}, \mathbb{R})$ was obtained by Sankaran and Uma [34]. In Ref. [20], the first author noticed that Brion's description also generalizes to the case of toric bundles (for more details, see Sect. 8.2). Like in the toric case, the description by Sankaran and Uma implicitly contains an algorithm to compute products of cohomology classes in the top degree. In Sect. 3.1, we provide further details.

We complement the descriptions of the cohomology ring of toric bundles with a generalization of the virtual polytope description. Like in the toric case, our description relies on (a generalization of) the BKK theorem (see Theorem 4.1). More precisely,

we reduce the computation of intersection numbers on the toric bundle to intersection numbers on the base. This provides a "BKK-type theorem" for any choice of a cohomology class in the base $\gamma \in H^*(B, \mathbb{R})$.

An equivalent description of the generalized BKK Theorem can be given as follows. Suppose the torus *T* has rank *n* and the real dimension of *B* is *k*. Similar to the toric case, a virtual polytope Δ defines a cohomology class $\rho(\Delta) \in H^2(E_{\Sigma}, \mathbb{R})$ on the *T*-toric bundle $p: E_{\Sigma} \to B$. In Theorem 4.3, for any given *j*, we define a map which associates with a virtual polytope Δ a cohomology class $\gamma_{2j}(\Delta) \in H^{2j}(B, \mathbb{R})$, such that

$$\rho(\Delta)^{n+j} \cdot p^*(\gamma) = \gamma_{2j}(\Delta) \cdot \gamma,$$

for any $\gamma \in H^{k-2j}(B, \mathbb{R})$. Here, "·" denotes the cup product on the respective cohomology ring. We call the class $\gamma_{2j}(\Delta)$ *the horizontal part* of $\rho(\Delta)^{n+j}$. We refer to Sect. 4 for further details.

In particular, our version of the BKK theorem allows us to compute the selfintersection polynomial on $H^2(E_{\Sigma}, \mathbb{R})$ (see Theorem 9.2), that is the map

$$H^2(E_{\Sigma}, \mathbb{R}) \to \mathbb{R}, \quad \gamma \mapsto \int_{E_{\Sigma}} \gamma^{n+k/2} \text{ if } k \text{ is even and zero otherwise.}$$

It should be mentioned that our statements above are true not only in the algebraic category, but more generally hold for smooth manifolds.

Furthermore, unlike the toric case or the case studied in Ref. [21], the cohomology ring of E_{Σ} is not generated in degree 2 in general. However, graded-commutative algebras which satisfy Poincaré duality accept a description generalizing the one used by Pukhlikov and the second author (see Theorem 7.16). In the case of commutative algebras satisfying Poincaré duality, an alternative description was later obtained in Ref. [27].

From the generalized BKK theorem and our description of graded commutative algebras with Poincaré duality, we obtain a description of the cohomology ring of toric bundles (see Theorem 8.1). In Sect. 8.1, we show that our approach naturally yields an alternative description of the cohomology ring of toric bundles obtained by Uma and Sankaran [34].

Our description is well suited to compute cohomology rings of toric bundles over a fixed base manifold B. In particular, a computation of the ring of conditions of horospherical homogeneous spaces naturally follows (Corollary 8.4). Recall that the ring of conditions is an intersection ring for (not necessarily complete) homogeneous spaces. Furthermore, for a connected complex reductive group G, a homogeneous space G/H is called *horospherical* if H is a closed subgroup in G containing a maximal unipotent subgroup.

We conclude the manuscript by applying our results to the case of toric bundles over generalized flag varieties G/P (also known as toroidal horospherical varieties). In Theorem 10.4, we obtain a version of the Brion–Kazarnovskii theorem [7, 22] for our setting. In particular, we recover the results of [24] (and thus the results of [21]). We continue with the case of toric bundles over full flag varieties G/B for which we give

two descriptions of the cohomology rings and the ring of conditions (Theorems 10.6 and 10.13). The latter description agrees with a construction of Newton–Okounkov bodies from [23]. Indeed, we introduce the notion of *fibered virtual polytopes* which is the extension of the family of polytopes accepting a projection onto a polytope in the positive Weyl chamber with fibers being string polytopes (see Sects. 10.3 and 10.4).

Finally, we would like to mention that the methods developed in this paper were used in Ref. [26] to study the topology of quasitoric bundles. Similar to toric bundles, quasitoric bundles are fiber bundles with quasitoric manifold as a fiber. A quasitoric manifold is a topological analog of smooth projective toric varieties which were introduced by Davis and Januszkiewicz in Ref. [12]. The results of [26] use the theory of multi-polytopes developed by Hattori and Masuda in Ref. [19].

Organization of the Paper

Section 2 provides further details for the two descriptions of the cohomology ring of toric varieties mentioned above. Section 3 collects results about the cohomology ring of toric bundles and horospherical varieties. Section 4 contains the statement of the generalized BKK theorem. Section 5 provides further details about virtual polytopes and the related convex chains. Section 6 contains the proof of the generalized BKK theorem. Section 7 studies graded-commutative finite-dimensional algebras satisfying Poincaré duality. Section 8 describes the cohomology ring of toric bundles (in general) and computes the ring of conditions of horospherical homogeneous spaces. Section 9 discusses a calculation of the self-intersection polynomial on the second cohomology group of toric bundles yielding an easier description of the cohomology ring provided that it is generated in degree 2. Section 10 contains a version of Brion–Kazarnovskii theorem and computes the cohomology ring of certain classes of toric bundles including toric bundles over full flag varieties G/B and projective bundles.

2 The Cohomology Ring of a Toric Variety

We assume fundamental knowledge of toric geometry and refer to [11] for further details and references. For the reader's convenience, we give further details of the two descriptions of the cohomology ring of toric varieties mentioned in Sect. 1. We continue to use the notation from the Introduction. In particular, M denotes a lattice, i.e., a finitely generated free abelian group, i.e., $M \simeq \mathbb{Z}^n$, $N = \text{Hom}_{\mathbb{Z}}(M, \mathbb{Z})$ its dual lattice, and $\langle \cdot, \cdot \rangle \cdot M \times N \to \mathbb{Z}$ their dual pairing. Suppose that X_{Σ} is a smooth projective toric variety given by a smooth projective fan $\Sigma \subseteq N_{\mathbb{R}} := N \otimes \mathbb{R}$.

2.1 Intersection Products in the Stanley–Reisner Description

The Stanley–Reisner description of the cohomology ring yields an algorithm to compute products of cohomology classes in the top degree of $H^*(X_{\Sigma}, \mathbb{R})$: As R_{Σ} is generated in degree 1, it suffices to consider monomials $x_{i_1}^{k_1} \dots x_{i_t}^{k_t}$. Recall that the graded piece of top degree of R_{Σ} is one-dimensional and generated by $x_{i_1} \dots x_{i_n}$ for any collection i_1, \ldots, i_n of indices, such that the rays $\rho_{i_1}, \ldots, \rho_{i_n}$ generate a fulldimensional cone in Σ . Indeed, all such monomials $x_{i_1} \cdots x_{i_n}$ yield the same element in R_{Σ} . Therefore, the evaluation of a monomial $x_{i_1}^{k_1} \ldots x_{i_t}^{k_t}$ in the top degree amounts to expressing it as a linear combination of square-free monomials.

For a monomial $x_{i_1}^{k_1} \cdots x_{i_t}^{k_t}$, let $\sum_{i=1}^t (k_i - 1)$ be its *multiplicity*, so that being square free is equivalent to having multiplicity 0. To simplify notation, consider the monomial $x_1^{k_1} \cdots x_t^{k_t}$ (always possible by reordering the variables). Suppose $x_1^{k_1} \cdots x_t^{k_t}$ is a monomial with multiplicity m > 0 and $k_1 > 1$. The goal is to express it as a linear combination of monomials of smaller multiplicity. If ρ_1, \ldots, ρ_t do not form a cone, then the monomial is equal to zero in R_{Σ} and we are done. Otherwise, the set $\{e_1, \ldots, e_t\}$ can be extended to a lattice basis of N, and thus, there is $\chi \in M$, such that $\chi(e_1) = 1$ and $\chi(e_j) = 0$, for $j = 2, \ldots, t$. Note that χ induces a linear relation in J, so that we obtain

$$x_1^{k_1} \dots x_t^{k_t} = x_1^{k_1-1} \dots x_t^{k_t} \cdot \left(-\sum_{k=t+1}^s \chi(e_k) x_k \right).$$

This is a linear combination of monomials of multiplicity m - 1. Applying this procedure inductively, we end up with a linear combination of square-free monomials, and thus obtain an evaluation of $x_1^{k_1} \cdots x_t^{k_t}$.

2.2 The Virtual Polytope Description

As noted in the introduction, an arbitrary line bundle on X_{Σ} can be expressed as a difference of ample line bundles which, by the toric dictionary, correspond to polytopes whose normal fan coarsens Σ . Recall that a fan Σ' in $N_{\mathbb{R}}$ coarsens Σ if every cone of Σ is contained in a cone of Σ' and $|\Sigma'| = |\Sigma|$. One also says that Σ refines Σ' . In particular, every cone of Σ' is a union of cones of Σ . One of the crucial ideas in Ref. [33] is to reformulate this observation on line bundles into convex geometry. Therefore, let us recall the definition of the space of virtual polytopes.

Let \mathcal{P}_{Σ}^+ be the set of polytopes in $M_{\mathbb{R}}:=M\otimes\mathbb{R}$, such that Σ refines their normal fans. Recall that the *Minkowski sum* of two polytope P, Q in $M_{\mathbb{R}}$ is given by

$$P + Q = \{p + q \colon p \in P, q \in Q\},\$$

and that the normal fan of a Minkowski sum is the coarsest common refinement of the individual fans (see [37, Proposition 7.12]). Thus, the Minkowski sum endows the structure of a monoid on \mathcal{P}_{Σ}^+ . It is well known that a commutative semigroup can be embedded in a group if the operation is *cancellative*, i.e., $P_1 + Q = P_2 + Q$ implies $P_1 = P_2$. It is straightforward to show that the Minkowski sum satisfies this property. We denote the smallest group \mathcal{P}_{Σ} containing the monoid \mathcal{P}_{Σ}^+ the group of *virtual polytopes* (associated to the fan Σ). Clearly, \mathcal{P}_{Σ}^+ accepts a natural multiplication map by the non-negative reals $\mathbb{R}_{\geq 0}$ which straightforwardly extends to a scalar multiplication by \mathbb{R} on the group of virtual polytopes, thus turning \mathcal{P}_{Σ} into a real vector space. Moreover, note that this also shows that \mathcal{P}_{Σ}^+ is a cone in \mathcal{P}_{Σ} , and as it generates the ambient vector space, it is full dimensional.

For us, the following alternative description of this vector space will be useful. Recall that one associates with a non-empty compact convex set $A \subseteq M_{\mathbb{R}}$ its *support* function

$$h_A: N_{\mathbb{R}} \to \mathbb{R}; \quad x \mapsto h_A(x):= \sup\{\langle a, x \rangle : a \in A\}.$$

In other words, given an exterior normal vector $x \neq 0$, the position of the corresponding supporting hyperplane is determined by its support function h_A , namely $A \subseteq \{y \in M_{\mathbb{R}} : \langle y, x \rangle \leq h_A(u)\}$ with the hyperplane $\{y \in M_{\mathbb{R}} : \langle y, x \rangle = h_A(u)\}$ intersecting A in a non-empty face. The crucial observation is that h_A uniquely describes A. Indeed, one can show that $A = \{x \in M_{\mathbb{R}} : \langle x, u \rangle \leq h_A(u) \text{ for all } u \in N_{\mathbb{R}}\}.$

Using the language of support functions, \mathcal{P}_{Σ}^+ and \mathcal{P}_{Σ} can be reformulated as follows:

$$\mathcal{P}_{\Sigma}^{+} = \{h \colon N_{\mathbb{R}} \to \mathbb{R} : \text{ convex, piecewise linear function on } \Sigma \},\$$
$$\mathcal{P}_{\Sigma} = \{h \colon N_{\mathbb{R}} \to \mathbb{R} : \text{ piecewise linear function on } \Sigma \},\$$

where a function $h: N_{\mathbb{R}} \to \mathbb{R}$ is piecewise linear on Σ if $h_{\sigma}:=h|_{\sigma}$ can be extended to a linear function on span $\{\sigma\}$ for any $\sigma \in \Sigma$. In addition, such a function is *convex* if $h(u) \ge h_{\sigma}(u)$ for any $\sigma \in \Sigma$ and $u \in \text{span}\{\sigma\}$.

With the above observation on line bundles, we have that $\mathcal{P}_{\Sigma}/M_{\mathbb{R}} \simeq H^2(X_{\Sigma}, \mathbb{R})$ where $M_{\mathbb{R}}$ is interpreted as a linear subspace of \mathcal{P}_{Σ} , namely the subspace of linear functions. Indeed, any $m \in M_{\mathbb{R}}$ can be expressed as a difference (P+m) - P for any polytope $P \in \mathcal{P}_{\Sigma}^+$. Note that the volume yields a well-defined polynomial function on $\mathcal{P}_{\Sigma}/M_{\mathbb{R}}$. The following theorem is implicit in Ref. [33] and forms a key idea to get a description of the cohomology ring $H^*(X_{\Sigma}, \mathbb{R})$. The following version is taken from [21, Theorem 1.1].

Theorem 2.1 Suppose $A = \bigoplus_{i=0}^{n} A_i$ is a graded finite-dimensional commutative algebra over a field \mathbb{K} of characteristic 0, such that A is generated (as an algebra) by the elements A_1 of degree one, $A_0 \simeq A_n \simeq \mathbb{K}$, and the bilinear map $A_i \times A_{n-i} \to A_n$ is non-degenerate for any i = 0, ..., n (Poincaré duality). Then

$$A \simeq \mathbb{K}[t_1, \dots, t_r] / \{ p(t_1, \dots, t_r) \in \mathbb{K}[t_1, \dots, t_r] :$$

$$p(\frac{\partial}{\partial x_1}, \dots, \frac{\partial}{\partial x_r}) f(x_1, \dots, x_r) = 0 \},$$

where we identify A_1 with \mathbb{K}^r via a basis v_1, \ldots, v_r and define $f : A_1 \simeq \mathbb{K}^r \to \mathbb{K}$ as the polynomial given by $f(x_1, \ldots, x_r) = (x_1v_1 + \cdots + x_rv_r)^n \in A_n \simeq k$.

Remark 2.2 Instead of identifying A_1 with \mathbb{K}^r , the previous theorem accepts a basisfree formulation in terms of Diff (A_1) , the ring of differential operators with constant coefficients: $A \simeq \text{Diff}(A_1)/\{p \in \text{Diff}(A_1): p(f) = 0\}$.

In Sect. 7, we study algebras with Poincaré duality which are not necessarily generated in degree 1. In particular, we prove Theorem 7.16 which is a generalization of Theorem 2.1. Then, in Theorem 9.3, we show that Theorem 2.1 is indeed a corollary of Theorem 7.16. 178

Under the identification $\mathcal{P}_{\Sigma}/M_{\mathbb{R}} \simeq H^2(X_{\Sigma}, \mathbb{R})$ together with BKK theorem, we get that the function $H^2(X_{\Sigma}, \mathbb{R}) \to H^{2n}(X_{\Sigma}, \mathbb{R}) \simeq \mathbb{R}; x \mapsto x^n$ corresponds to the volume polynomial on \mathcal{P}_{Σ}/M , and thus, we obtain (using the formulation from Remark 2.2)

$$H^*(X_{\Sigma}, \mathbb{R}) \simeq \operatorname{Diff}(\mathcal{P}_{\Sigma}/M_{\mathbb{R}})/\operatorname{Ann}(\operatorname{Vol}) \simeq \operatorname{Diff}(\mathcal{P}_{\Sigma})/\operatorname{Ann}(\operatorname{Vol}),$$

where Ann(Vol) denotes the ideal of differential operators which annihilate the volume polynomial Vol: $\mathcal{P}_{\Sigma} \to \mathbb{R}$.

3 Toric Bundles and Horospherical Varieties

In this section, we collect several facts on toric bundles which will be used below. We also introduce horospherical varieties which yield an important class of toric bundles.

3.1 Toric Bundles

Let *G* be a topological group and let $p: E \to B$ be a *G*-principal bundle over a topological space *B*, i.e., $p: E \to B$ is a fiber bundle with structure group *G* equipped with a *G*-atlas, such that the action of *G* on the fibers $p^{-1}(b)$ for any $b \in B$ is free and transitive. In other words, $p: E \to B$ can be characterized as a *G*-torsor, where each fiber of *p* is isomorphic to *G*, although there is no canonical identification of the fibers with the group *G*. To any *G*-principal bundle $p: E \to B$ and any topological space *X* equipped with a continuous action by *G*, one associates a fiber bundle by introducing a (right) action on the product $E \times X$

$$(e, x) \cdot g := (e \cdot g, g^{-1} \cdot x).$$

The *associated fiber bundle* is given as the quotient $E \times_G X := (E \times X)/G$. It is a fiber bundle with fiber X and base B. If G = T is an algebraic torus, then a T-principal bundle is also called a *torus bundle*. If, furthermore, X is a T-toric variety, then the associated fiber bundle $E_X := E \times_T X$ is said to be a *toric bundle*.

Crucial to the understanding of the cohomology of fiber bundles is the following theorem (see, for instance, [9, Theorem 5.11] or [18, Theorem 4D.1]):

Theorem 3.1 (Leray–Hirsch) Let E be a fiber bundle with fiber F over a compact manifold B. If there are global cohomology classes u_1, \ldots, u_r on E whose restrictions $i^*(u_i)$ form a basis for the cohomology of each fiber F (where $i: F \rightarrow E$ is the inclusion), then we have an isomorphism of vector spaces

$$H^*(B,\mathbb{R})\otimes H^*(F,\mathbb{R})\to H^*(E,\mathbb{R}); \quad \sum_{i,j}b_i\otimes i^*(u_j)\mapsto \sum_{i,j}p^*(b_i)\cdot u_j.$$

Corollary 3.2 If T is an algebraic torus, $p: E \rightarrow B$ a T-toric bundle as in Theorem 3.1, and X a smooth projective T-toric variety, then as a group the cohomology

of $E_X = E \times_T X$ is given by

$$H^*(E_X, \mathbb{R}) \simeq H^*(B, \mathbb{R}) \otimes H^*(X, \mathbb{R}).$$

Proof As a group, the cohomology of smooth projective toric varieties is generated by classes Poincaré dual to orbit closures. For any orbit closure \overline{O} of X, let $E_{\overline{O}} = E \times_T \overline{O}$ be a submanifold of E_X . Note that E_X is compact, and thus, Poincaré duality applies. The cohomology classes u_1, \ldots, u_r Poincaré dual to these submanifolds satisfy the condition of Theorem 3.1. The statement follows.

Remark 3.3 In Corollary 3.2, we rely on the fact that the monodromy group of a toric bundle (in the sense as defined above) does not permute the orbits of the fiber toric variety. It is worth noting that the more general class of fibrations with non-trivial orbit monodromy is beyond the scope of this paper.

Corollary 3.2 yields a description of the cohomology group of E_X . Crucial for this description is the map which associates with any orbit closure \overline{O} of X the Poincaré dual of the corresponding T-invariant submanifold of E_X . By restricting this map to divisors and extending using linearity, we obtain

 $\rho: \mathcal{P}_{\Sigma} \to H^2(E_X, \mathbb{R}) \quad \text{(where } \Sigma \text{ is the fan of } X = X_{\Sigma}\text{)},$

which plays an important role in our description of the cohomology ring of E_X . We provide more details for ρ . Recall that ρ_1, \ldots, ρ_r denote the rays of Σ with integer generators e_1, \ldots, e_r . Let D_1, \ldots, D_r be the corresponding divisors in X. We also write D_i for the submanifold $E \times_T D_i$ of E_X . Then, for $\Delta \in \mathcal{P}_{\Sigma}$, we have

$$\rho(\Delta) = \sum_{i=1}^{\prime} h_{\Delta}(e_i)[D_i] \in H^2(E_X, \mathbb{R}),$$

where $[D_i]$ is the class Poincaré dual to $D_i \subseteq E_X$ and $h_\Delta \colon N_{\mathbb{R}} \to \mathbb{R}$ is the support function of Δ .

The following observation about $\rho(\cdot)$ will be crucial for our approach. Any character $\lambda \in M$ defines a one-dimensional representation \mathbb{C}_{λ} of T, namely $t \cdot z = \lambda(t)z$ for $t \in T$, and $z \in \mathbb{C}_{\lambda}$. If \mathcal{L}_{λ} denotes the associated complex line bundle on B, i.e., $\mathcal{L}_{\lambda} \simeq E \times_T \mathbb{C}_{\lambda}$, then $\mathcal{L}_{\lambda+\mu} = \mathcal{L}_{\lambda} \otimes \mathcal{L}_{\mu}$, and thus, we obtain a group homomorphism

$$c: M \to H^2(B, \mathbb{Z}), \quad \lambda \mapsto c_1(\mathcal{L}_{\lambda}),$$

where $c_1(\mathcal{L}_{\lambda})$ is the first Chern class. By linearity, we extend the homomorphism to a map of vector spaces

$$c\colon M_{\mathbb{R}}\to H^2(B,\mathbb{R}).$$

Example 3.4 Consider the torus bundle $p: (\mathbb{C}^2 \setminus \{0\}) \to \mathbb{P}^1$ where $T = \mathbb{C}^*$ acts diagonally on $\mathbb{C}^2 \setminus \{0\}$, i.e., $t \cdot x = (t^{-1}x_0, t^{-1}x_1)$ for $t \in \mathbb{C}^*$ and $x = (x_0, x_1) \in \mathbb{C}^2 \setminus \{0\}$.

Let $k \in M_T = \mathbb{Z}$. Then, the *T*-action on \mathbb{C}_k is given by $t \cdot z = t^k z$. For k = 1, we get that *T* acts on $(\mathbb{C}^2 \setminus \{0\}) \times \mathbb{C}_1$ via $t \cdot ((x_0, x_1), z) = ((t^{-1}x_0, t^{-1}x_1), tz)$.

Notice that the following map induces an isomorphism between \mathcal{L}_1 and the tautological bundle on \mathbb{P}^1 :

 $(\mathbb{C}^2 \setminus \{0\}) \times \mathbb{C}_1 \to \mathbb{P}^1 \times \mathbb{C}^2; \quad ((x_0, x_1), z) \mapsto ([x_0 : x_1], (zx_0, zx_1)).$

This follows from the fact that this map factors through the quotient $(\mathbb{C}^2 \setminus \{0\}) \times_{\mathbb{C}^*} \mathbb{C}_1 = ((\mathbb{C}^2 \setminus \{0\}) \times \mathbb{C}_1) / \mathbb{C}^*$. Hence, we get that $\mathcal{L}_1 = \mathcal{O}_{\mathbb{P}^1}(-1)$, and thus $c_1(\mathcal{L}_1) = -1 \in H^2(\mathbb{P}^1, \mathbb{Z}) = \mathbb{Z}$. More generally, for $k \in M_T = \mathbb{Z}$, we get $c(k) = -k \in H^2(\mathbb{P}^1, \mathbb{Z})$.

It is worth noticing that the *T*-action on \mathbb{C}_{λ} identifies $\mathbb{C}_{\lambda} \setminus \{0\}$ with $T/\ker(\lambda)$. Since the fibers of \mathcal{L}_{λ} can be identified with \mathbb{C}_{λ} , we can view them in a similar way as $T/\ker(\lambda)$.

Proposition 3.5 Let Σ be a smooth complete fan, and $p: E_X \to B$ as before. Then, for any character $\lambda \in M$

$$p^*c(\lambda) = \rho(\lambda),$$

where on the right-hand side of the equality λ is regarded as a virtual polytope.

Proof. A character λ of T defines a T-invariant divisor div (λ) on X which can be expressed as div $(\lambda) = \sum_{i=1}^{r} k_i D_i$ where the $D_i \subseteq X$ are distinct irreducible divisors and $k_i \in \mathbb{Z}$. Every divisor D_i comes with a (naturally) linearized line bundle \mathcal{L}_i and a global regular section s_i , such that div $(s_i) = D_i$. The sections s_i yield generic global (smooth) sections of the associated line bundle $E \times_T \mathcal{L}_i \to E \times_T X$ with degeneracy locus $E \times_T D_i$. By [16, Gauss–Bonnet Formula II, p. 413], $c_1(E \times_T \mathcal{L}_i)$ is Poincaré dual to $E \times_T D_i$. Using linearity of c_1 , we get $c_1(E \times_T \mathcal{L}_\lambda) = \rho(\lambda)$. It is straightforward to verify that $E \times_T \mathcal{L}_\lambda = p^* \mathcal{L}_\lambda$, and so

$$p^*c(\lambda) = c_1(p^*\mathcal{L}_{\lambda}) = c_1(E \times_T \mathcal{L}_{\lambda}) = \rho(\lambda).$$

The map of lattices $c: M \to H^2(B, \mathbb{Z})$ is an invariant that uniquely describes torus bundles in the topological category. Indeed, a torus bundle *E* can be recovered from $c: M \to H^2(B, \mathbb{Z})$ as follows.

Choose a basis e_1, \ldots, e_k of M. Since (topological) complex line bundles are classified by their Chern classes, there exist bundles \mathcal{L}_i on B uniquely determined by $c_1(\mathcal{L}_i) = c(e_i) \in H^2(B, \mathbb{Z})$ for $i = 1, \ldots, k$. The rest of the construction follows a general recipe for how to get a torus bundle $E \to B$ from a family of (topological) complex line bundles $\mathcal{L}_1, \ldots, \mathcal{L}_k$ on B:

Consider the vector bundle $\mathcal{L}_1 \oplus \ldots \oplus \mathcal{L}_k$ of rank *k* over *B*. Note for any subset $\{i_1, \ldots, i_s\} \subseteq \{1, \ldots, k\}$, the vector bundle $\mathcal{L}_{i_1} \oplus \ldots \oplus \mathcal{L}_{i_s}$ of rank s < k naturally embeds into $\mathcal{L}_1 \oplus \ldots \oplus \mathcal{L}_k$. Let $E \subseteq \mathcal{L}_1 \oplus \ldots \oplus \mathcal{L}_k$ be the complement of the coordinate vector subbundles $\mathcal{L}_{i_1} \oplus \ldots \oplus \mathcal{L}_{i_s}$. The algebraic torus $(\mathbb{C}^*)^k$ acts freely on

E via coordinate-wise scaling, and $E \rightarrow B$ is a (topological) principal torus bundle. We arrive at the following result.

Proposition 3.6 Let *B* be a closed oriented manifold and *T* be an algebraic torus with character lattice M(T). Then, (topological) *T*-principal bundles over *B* are in bijection with homomorphisms $c : M(T) \to H^2(B, \mathbb{Z})$.

Example 3.7 Let *C* be a smooth algebraic curve and let $\mathcal{L}_1, \ldots, \mathcal{L}_k$ be line bundles on *C* of degrees d_1, \ldots, d_k , respectively. Suppose $E \subset \mathcal{L}_1 \oplus \ldots \oplus \mathcal{L}_k$ is the principal $T = (\mathbb{C}^*)^k$ -bundle as before. Then, for a character $\lambda = (\lambda_1, \ldots, \lambda_k) \in M_T = \mathbb{Z}^k$, we have $\mathcal{L}_{\lambda} = \mathcal{L}_1^{\lambda_1} \otimes \ldots \otimes \mathcal{L}_k^{\lambda_k}$ and the homomorphism $c \colon M_T \to H^2(C, \mathbb{Z}) \simeq \mathbb{Z}$ is given by

$$c: (\lambda_1, \ldots, \lambda_k) \mapsto \lambda_1 d_1 + \cdots + \lambda_k d_k.$$

If the base manifold *B* is an algebraic variety, there is an analogous description of algebraic *T*-principal bundles over *B*. In this case, one has to consider the homomorphism $\tilde{c}: M(T) \rightarrow \text{Pic}(B)$ given by

$$\widetilde{c}: M(T) \to \operatorname{Pic}(B); \ \lambda \mapsto \mathcal{L}_{\lambda} \in \operatorname{Pic}(B).$$

We conclude this section with the following description of the cohomology ring of toric bundles by Sankaran and Uma. We note that this description is true in greater generally as stated here and we refer the reader to [34, Theorem 1.2] for the complete statement. The following version will be sufficient for what follows.

Theorem 3.8 We continue to use the notation from above. If X_{Σ} is smooth and projective, the cohomology ring $H^*(E_{\Sigma}, \mathbb{R})$ is isomorphic (as an $H^*(B, \mathbb{R})$ -algebra) to the quotient of $H^*(B, \mathbb{R})[x_1, \ldots, x_r]$ by

$$\left(\left\langle x_{j_1}\cdots x_{j_k}:\rho_{j_1},\ldots,\rho_{j_k}\text{ do not span a cone of }\Sigma\right\rangle+\left\langle c\left(\lambda\right)-\sum_{i=1}^n\langle e_i,\lambda\rangle x_i:\lambda\in M\right\rangle\right).$$

Note the similarities with the Stanley–Reisner description of the cohomology ring of toric varieties. Indeed, the first ideal in Theorem 3.8 corresponds to the Stanley–Reisner ideal of the corresponding toric variety. In particular, the algorithm from Sect. 2.1 can be used to compute products in the top degree of $H^*(E_X, \mathbb{R})$. In the end of Sect. 8.1, we will give an alternative proof of Theorem 3.8 based on our version of BKK Theorem.

3.2 Horospherical Varieties

In this section, we recall horospherical varieties which provide an important source of examples of toric bundles. We conclude with a description of their associated cohomology ring. Let *G* be a connected reductive complex algebraic group. A closed subgroup $H \subseteq G$ is called *horospherical* if it contains a maximal unipotent subgroup *U* of *G* and E:=G/H is called a *horospherical homogeneous space*. Let $B \subseteq G$ be a Borel subgroup whose unipotent radical is *U*. It is known that $P:=N_G(H)$ is a parabolic subgroup containing *B*. The natural quotient $p: E \rightarrow G/P$ is a torus bundle with fiber: the torus T = P/H.

For the reader's convenience, we illustrate these notions with the following classical example.

Example 3.9 Let $G = GL_n$ be the group of invertible $n \times n$ -matrices. A maximal unipontent subgroup U of GL_n is given by the group of upper triangular matrices with 1's on the diagonal. The normalizer $B = N_G(U)$ is equal to the group of invertible upper triangular matrices which is a Borel subgroup of GL_n . Then, B/U is isomorphic to $(\mathbb{C}^*)^n$ which can be identified with the diagonal subgroup of GL_n (a maximal torus in GL_n).

In this case, the homogeneous space GL_n/U is a principal $(\mathbb{C}^*)^n$ -bundle over GL_n/B where GL_n/B can be identified with the variety Fl_n of full flags in \mathbb{C}^n . The homogeneous space GL_n/U itself has a modular interpretation. Points of GL_n/U parameterize full flags $F = (F_0, \ldots, F_n)$ with a choice of a basis vector in F_i/F_{i-1} for $i = 1, \ldots, n$. Under this description of GL_n/U , the projection $p: \operatorname{GL}_n/U \to \operatorname{GL}_n/B \simeq \operatorname{Fl}_n$ is given by the forgetful morphism.

Let *M* be the character lattice of *T*. Let $\Sigma \subseteq N_{\mathbb{R}}$:=Hom_{\mathbb{Z}}(*M*, \mathbb{R}) be a fan with corresponding toric variety $X = X_{\Sigma}$. The associated toric bundle $p: E_X \to G/P$ is a horospherical variety. In general, an irreducible normal *G*-variety *Y* together with an open equivariant embedding $G/H \hookrightarrow Y$ is called a *horospherical variety*. Recall that a morphism of *G*-varieties $\varphi: Y \to Y'$ is called *equivariant* if $\varphi(g \cdot y) = g \cdot \varphi(y)$ for any $g \in G$ and $y \in Y$. We refer to [28, 35] for further details and references on horospherical varieties.

We conclude this section with a combinatorial description of the cohomology ring $H^*(E_X, \mathbb{R})$. Recall from the previous section that any character $\lambda \in M$ gives rise to a line bundle \mathcal{L}_{λ} on G/P, so that we obtain a group homomorphism $c: M \to H^2(G/P, \mathbb{Z})$. The following statement combinatorially describes the cohomology ring of E_X provided X_{Σ} is smooth and projective. It is a special case of a more general result.

Theorem 3.10 ([34, Theorem 1.2]) Suppose Σ has rays ρ_1, \ldots, ρ_r with primitive vectors $v_1, \ldots, v_n \in N$:=Hom_{\mathbb{Z}} (M, \mathbb{Z}) along the edges ρ_i . If X_{Σ} is smooth and projective, then the cohomology ring $H^*(E_X, \mathbb{R})$ is isomorphic as an $H^*(G/P, \mathbb{R})$ -algebra to the quotient of $H^*(G/P, \mathbb{R})[x_1, \ldots, x_r]$ by the sum of ideals

$$\langle x_{j_1}\cdots x_{j_k}: \rho_{j_1},\ldots,\rho_{j_k} \text{ do not span a cone of } \Sigma \rangle + \langle c(\lambda) - \sum_{i=1}^r \langle v_i,\lambda \rangle x_i:\lambda \in M \rangle.$$

4 BKK Theorems for Toric Bundles

This section provides an overview of the main results of this paper: Theorems 4.1 and 4.3. Details and proofs will be given in Sect. 6, and in Sect. 8, these theorems will be used to describe the cohomology ring of toric bundles.

As before, let $p: E \to B$ be a principal torus bundle with respect to a torus $T \simeq (\mathbb{C}^*)^n$ over a compact smooth orientable manifold B of real dimension k. Let M be the character lattice of T and $\Sigma \subseteq N_{\mathbb{R}} = \text{Hom}_{\mathbb{Z}}(M, \mathbb{R})$ be a smooth complete fan which gives rise to a toric variety $X = X_{\Sigma}$. Let E_X be the total space of the associated toric bundle. Note that E_X is a compact smooth orientable manifold of real dimension k + 2n. To keep notation simple, we denote the projection map of the toric bundle by $p: E_X \to B$ as well.

Our main theorems show that a choice of a natural number $i \leq \frac{k}{2}$ and $\gamma \in H^{k-2i}(B, \mathbb{R})$ gives rise to a BKK -type theorem. First, we define two functions I_{γ} and F_{γ} on \mathcal{P}_{Σ} as follows.

Let $f_{\gamma} \colon M_{\mathbb{R}} \to \mathbb{R}$ be given by

$$f_{\gamma}(x) = c(x)^{l} \cdot \gamma,$$

where "·" denotes the cup product of the cohomology ring $H^*(B, \mathbb{R})$. Here, we identified the top cohomology group $H^k(B, \mathbb{R})$ with \mathbb{R} by pairing with the fundamental class of *B*. Since $c: M_{\mathbb{R}} \to H^2(B, \mathbb{R})$ is a linear map, f_{γ} is a homogeneous polynomial of degree *i* on $M_{\mathbb{R}}$. This leads to the definition of I_{γ}

$$I_{\gamma} \colon \mathcal{P}_{\Sigma} \to \mathbb{R}; \quad I_{\gamma}(\Delta) \coloneqq \int_{\Delta} f_{\gamma}(x) \mathrm{d}\mu \quad \text{for } \Delta \in \mathcal{P}_{\Sigma}^{+},$$

where μ denotes the Lebesgue measure on $M_{\mathbb{R}}$ normalized with respect to the lattice M, i.e., a cube spanned by an affine lattice basis of M has volume 1. Note that we gave an explicit formula for I_{γ} on the full-dimensional cone \mathcal{P}_{Σ}^+ . By Theorem 5.5, I_{γ} extends to a homogeneous polynomial of degree n + i on \mathcal{P}_{Σ} .

Recall the definition of $\rho: \mathcal{P}_{\Sigma} \to H^2(E_X, \mathbb{R})$ from Sect. 3.1. This leads to the definition of the function F_{γ}

$$F_{\gamma}: \mathcal{P}_{\Sigma} \to \mathbb{R}; \quad F_{\gamma}(\Delta) := \rho(\Delta)^{n+\iota} \cdot p^*(\gamma).$$

Clearly, F_{ν} is a homogeneous polynomial of degree n + i on \mathcal{P}_{Σ} .

The main result of this section is the following analog of the BKK theorem for toric bundles. Indeed, it expresses certain intersection numbers of cohomology classes as mixed integrals.

Theorem 4.1 *The polynomials* I_{γ} *and* F_{γ} *are proportional with coefficient of proportionality given by*

$$(n+i)! \cdot I_{\gamma}(\Delta) = i! \cdot F_{\gamma}(\Delta)$$
 for any $\Delta \in \mathcal{P}_{\Sigma}$.

In particular, the polarizations of I_{γ} and F_{γ} are proportional multilinear forms, i.e., for any $\Delta_1, \ldots, \Delta_{n+i} \in \mathcal{P}_{\Sigma}$

$$(n+i)! \cdot I_{\gamma}(\Delta_1, \ldots, \Delta_{n+i}) = i! \cdot F_{\gamma}(\Delta_1, \ldots, \Delta_{n+i}).$$

For the reader's convenience, we recall the concept of polarization (or equivalently mixed integrals). Let V be a (possibly infinite-dimensional) vector space. Here, $V = P_{\Sigma}$ is finite-dimensional; however, eventually, we want to consider \mathcal{P} which is an infinite-dimensional vector space.

Definition 4.2 The *Lie derivative* $L_v f(x)$ of a function $f: V \to \mathbb{R}$ with respect to a vector $v \in V$ at a point $x \in V$ is the limit

$$L_v f(x) = \lim_{t \to 0} \frac{f(x + t \cdot v) - f(x)}{t},$$

provided that this limit exists.

Note that for a finite-dimensional vector space V, the concept of polynomials $f: V \to \mathbb{R}$ is well known. In the next section, we recall how to extend the concept of polynomials to the infinite-dimensional case. As the concept of polarization works both in the finite- and the infinite-dimensional case, we provide details for the general situation here. The careful reader may want to consult Definition 5.1 before reading further.

Recall that for a homogeneous polynomial $f: V \to \mathbb{R}$ of degree *m*, the *polarization* of *f* is the unique symmetric multilinear form $g: V^m \to \mathbb{R}$, such that $g(v, \ldots, v) = f(v)$. It is a well-known fact that for any vector space *V* and any homogeneous polynomial *f* of degree *m*, the polarization exists and can be defined as follows:

$$g(v_1, \dots, v_m) = \frac{1}{m!} L_{v_1} \dots L_{v_m} f.$$
 (1)

For a homogeneous polynomial $f: M_{\mathbb{R}} \to \mathbb{R}$, the polarization of I_f is called the *mixed integral of* f.

Note that in the case of a trivial torus bundle $p: T \to pt$, Theorem 4.1 reduces to the classical BKK theorem. We conclude this section with an alternative interpretation of Theorem 4.1 which can be favorable for certain applications. By the Leray–Hirsch Theorem (see Corollary 3.2), $H^*(E_X, \mathbb{R}) \simeq H^*(B, \mathbb{R}) \otimes H^*(X, \mathbb{R})$, and so, for $\Delta \in \mathcal{P}_{\Sigma}$, the cycle $\rho(\Delta)^{n+i} \in H^{2n+2i}(E_X, \mathbb{R})$ can be written as

$$\rho(\Delta)^{n+i} = b_{2n+2i} \otimes x_0 + b_{2n+2i-2} \otimes x_2 + \dots + b_{2i+2} \otimes x_{2n-2} + b_{2i} \otimes x_{2n},$$

with $b_s \in H^s(B, \mathbb{R})$ and $x_r \in H^r(X, \mathbb{R})$. As X is a smooth, complete toric variety, its cohomology groups in odd degrees vanish, and therefore, $x_{2k+1} = 0$ for any k. If x_{2n} is normalized, such that it is dual to a point, we call b_{2i} the *horizontal part* of $\rho(\Delta)^{n+i}$. Equivalently, the horizontal part b_{2i} of $\rho(\Delta)^{n+i}$ is the unique class in $H^{2i}(B, \mathbb{R})$, such

that

$$\rho(\Delta)^{n+\iota} \cdot p^*(\eta) = b_{2i} \cdot \eta,$$

for any $\eta \in H^{k-2i}(B, \mathbb{R})$. Then, Theorem 4.1 accepts the following reformulation. **Theorem 4.3** For any $\Delta \in \mathcal{P}_{\Sigma}$, the horizontal part of $\rho(\Delta)^{n+i}$ can be computed as

$$b_{2i} = \frac{(n+i)!}{i!} \int_{\Delta} c(x)^i \mathrm{d}x.$$

Note that $c(\cdot)^i \colon M_{\mathbb{R}} \to H^{2i}(B, \mathbb{R})$ is a vector valued map whose components (after choosing suitable coordinates) are given by homogeneous polynomials of degree *i*. Thus, the integral in Theorem 4.3 exists. Furthermore, although we show that Theorem 4.1 implies Theorem 4.3, in fact, they are equivalent.

Proof Since $H^*(B, \mathbb{R})$ satisfies Poincaré duality, it suffices to check that for any $\gamma \in H^{k-2i}(B, \mathbb{R})$, we have

$$i! \cdot \gamma \cdot b_{2i} = (n+i)! \cdot \gamma \cdot \int_{\Delta} c(x)^i \mathrm{d}x.$$

Recall from Theorem 3.1 that there are $u_1, \ldots, u_r \in H^*(E_X, \mathbb{R})$, such that the restrictions $i^*(u_i)$ form a basis for the cohomology of each fibre X where $i: X \to E_X$ is the inclusion. In particular, there are $y_i \in \mathbb{R}u_1 \oplus \ldots \oplus \mathbb{R}u_r$, such that $i^*(y_i) = x_i$. Since x_{2n} is Poincaré dual to a point (say to a torus fixed point $x \in X$), it follows that $y_{2n} = E \times_T \{x\}$, i.e., $y_{2n} = [S]^*$ is the class Poincaré dual to a section of p.

Let $[pt]^* \in H^k(B, \mathbb{R})$ be the class dual to a point in B. For $\gamma \in H^{k-2i}(B, \mathbb{R})$, let $\gamma \cdot b_{2i} = a \cdot [pt]^*$ for some real number $a \in \mathbb{R}$. Then

$$p^{*}(\gamma) \cdot \rho(\Delta)^{n+i} = p^{*}(\gamma) \cdot (p^{*}(b_{2n+2i}) \cdot y_{0} + p^{*}(b_{2n+2i-2}) \cdot y_{2} + \cdots + p^{*}(b_{2i+2}) \cdot y_{2n-2} + p^{*}(b_{2i}) \cdot y_{2n}) = p^{*}(\gamma \cdot b_{2i}) \cdot y_{2n} = a \cdot p^{*}([\text{pt}]^{*}) \cdot E \times_{T} \{x\} = a \cdot [X]^{*} \cdot [S]^{*} = a \in H^{2n+k}(E_{X}, \mathbb{R}) \simeq \mathbb{R},$$

where $[X]^*$ is the class dual to a fibre of p. Hence, if we interpret $\gamma \cdot b_{2i}$ and $p^*(\gamma) \cdot \rho(\Delta)^{n+i}$ as real numbers (multiples of classes dual to a point), we get

$$\gamma \cdot b_{2i} = p^*(\gamma) \cdot \rho(\Delta)^{n+i} = F_{\gamma}(\Delta).$$

On the other hand

$$\gamma \cdot \int_{\Delta} c(x)^i \mathrm{d}x = \int_{\Delta} \gamma \cdot c(x)^i \mathrm{d}x = I_{\gamma}(\Delta).$$

The statement follows by Theorem 4.1.

Like Theorem 4.1, Theorem 4.3 admits a polarized version.

🖉 Springer

5 Convex Chains

In this subsection, we recall how to extend the definition of $I_{\gamma} : \mathcal{P}_{\Sigma}^+ \to \mathbb{R}$ to all of \mathcal{P}_{Σ} . In exposition we mostly follow [32]. Suppose that $\Sigma \subseteq N_{\mathbb{R}}$ is a *simplicial* fan, i.e., any cone in Σ is spanned by part of a basis of $N_{\mathbb{R}}$. Then, the primitive generators e_1, \ldots, e_s of the rays $\Sigma(1) = \{\rho_i = \mathbb{R}_{\geq 0}e_i : i = 1, \ldots, s\}$ naturally induce coordinates on \mathcal{P}_{Σ} : a piecewise linear function $h \in \mathcal{P}_{\Sigma}$ is uniquely determined by the tuple $(h(e_1), \ldots, h(e_s))$. We denote the corresponding coordinates on $\mathcal{P}_{\Sigma} \simeq \mathbb{R}^s$ by (h_1, \ldots, h_s) .

We want to work with the space of virtual polytopes \mathcal{P} . This is an infinitedimensional vector space obtained as inverse limit $\mathcal{P} = \varinjlim \mathcal{P}_{\Sigma}$ where the limit is taken over all complete fans Σ (partially ordered with respect to refinement of fans). As mentioned above, we consider polynomials, like I_{γ} , on this space. Let us recall the following notion of polynomials on a (possibly infinite-dimensional) real vector space V:

On the vector space of functions $g: V \to \mathbb{R}$, introduce the *(forward) difference operator*

$$D_v g(x) := g(x + v) - g(x)$$
 where $v \in V$.

Recall that a real topological vector space U is a real vector space equipped with a topology that makes the vector space operations $+: U \times U \rightarrow U$ and $\cdot: \mathbb{R} \times U \rightarrow U$ continuous. It is a known fact that every finite-dimensional Hausdorff topological vector space has the usual topology. From now on, we consider all finite-dimensional subspaces $U \subseteq V$ to be equipped with a topology, such that U is a Hausdorff topological vector space.

Definition 5.1 A function $f: V \to \mathbb{R}$ is called a *polynomial of degree* $\leq m$ if for any $v_1, \ldots, v_{m+1} \in V$

$$D_{v_1} \cdots D_{v_{m+1}} f(x) = 0 \quad \text{for any } x \in V,$$

and the restriction $f|_U : U \to \mathbb{R}$ is continuous for any finite-dimensional vector space $U \subseteq V$.

Remark 5.2 It is straightforward to show that, if $f: V \to \mathbb{R}$ is a polynomial and $v_1 \dots, v_n \in V$ are linearly independent, then for any $\lambda_1, \dots, \lambda_n \in \mathbb{R}$, we have

$$f(\lambda_1 v_1 + \dots + \lambda_n v_n) = \sum_{\alpha \in \mathbb{Z}_{\geq 0}^n} D^{\alpha} f(0) \binom{\lambda_1}{\alpha_1} \cdots \binom{\lambda_n}{\alpha_n},$$
(2)

where $D^{\alpha} := D_{v_1}^{\alpha_1} \cdots D_{v_n}^{\alpha_n}$ and $\binom{\lambda}{\alpha} = \frac{\lambda \cdot (\lambda - 1) \cdots (\lambda - \alpha + 1)}{\alpha!}$ denotes the binomial coefficient. Note that Definition 5.1 deviates from [32, Section 2, Definition 1] by the additional assumption of continuity. Indeed, without this assumption, equation (2) would be true for $\lambda_1, \ldots, \lambda_n \in \mathbb{Q}$ (which is sufficient in Ref. [32]). However, in our setting it is more natural to work over the field of real numbers, which makes this further assumption necessary.

In view of Remark 5.2, we may rephrase Definition 5.1 as follows:

Proposition 5.3 A function $f: V \to \mathbb{R}$ is a polynomial of degree $\leq m$ if and only if its restriction to any finite-dimensional subspace $U \subseteq V$ is a polynomial of degree $\leq m$ in the usual sense, i.e., an element in Sym (U^*) .

Recall the objective of this section: Any smooth function $f: M_{\mathbb{R}} \to \mathbb{R}$ defines a function I_f on the cone of convex polytopes via (here, μ denotes the standard Lebesgue measure on $M_{\mathbb{R}}$)

$$I_f: \mathcal{P}^+ \to \mathbb{R}; I_f(\Delta) = \int_{\Delta} f(x) \mathrm{d}\mu.$$

Our goal is to extend I_f to the whole space of virtual polytopes \mathcal{P} . We recall how this is done following [32].

Observe that $I_f: \mathcal{P}^+ \to \mathbb{R}$ is a *finitely additive measure* (also called a *valuation*) on the cone of convex polytopes, i.e., for any two convex polytopes $P, Q \in \mathcal{P}^+$, such that $P \cup Q$ is in \mathcal{P}^+ , we have

$$I_f(P \cup Q) = I_f(P) + I_f(Q) - I_f(P \cap Q).$$

A fundamental idea in Ref. [32] is the introduction of the group of *convex chains* $Z(M_{\mathbb{R}})$, i.e., the additive group consisting of functions of the form $\alpha = \sum_{i=1}^{k} n_i \mathbb{1}_{P_i}$ for integers n_i and convex polytopes $P_i \in \mathcal{P}^+$, where $\mathbb{1}_{P_i}$ denotes the characteristic function of the set P_i . The *degree* of a convex chain $\alpha \in Z(M_{\mathbb{R}})$ is defined to be $\sum_{i=1}^{k} n_i$ if $\alpha = \sum_{i=1}^{k} n_i \mathbb{1}_{P_i}$ for some $P_i \in \mathcal{P}^+$ (this is a well-defined number by [32, Proposition-Definition 2.1]). In Ref. [32, Proposition-Definition 2.3], Pukhlikov and the second author observe that the Minkowski sum of polytopes induces a ring structure on $Z(M_{\mathbb{R}})$ via

$$\mathbb{1}_{P} \star \mathbb{1}_{Q} = \mathbb{1}_{P+Q}$$
 for $P, Q \in \mathcal{P}^{+}$.

Clearly, the semigroup \mathcal{P}^+ can be embedded in the algebra of convex chains $Z(M_{\mathbb{R}})$. Furthermore, in Ref. [32, Section 6], it is shown that \mathcal{P} can be identified with the set of invertible convex chains of degree 1.

Observe that finitely additive measures (as defined above) are in correspondence with homomorphisms of additive groups $\varphi \colon Z(M_{\mathbb{R}}) \to \mathbb{R}$. A finitely additive measure is *polynomial of degree* $\leq m$ if, for each $\alpha \in Z(M_{\mathbb{R}})$, the function $M_{\mathbb{R}} \to \mathbb{R}$ given by $\lambda \mapsto \varphi(\tau(\lambda, \alpha))$ is polynomial of degree $\leq m$ where $\tau(\lambda, \cdot) \colon Z(M_{\mathbb{R}}) \to Z(M_{\mathbb{R}})$ denotes the linear map induced by translation by λ , i.e., $\tau(\lambda, \alpha)(x) = \alpha(x - \lambda)$. A fundamental result about polynomial finitely additive measures is the following theorem:

Theorem 5.4 [32, Corollary 7.5] *A polynomial finitely additive measure* $\varphi \colon Z(M_{\mathbb{R}}) \to \mathbb{R}$ *of degree* $\leq k$ *restricted to the space of virtual polytopes* $\varphi \colon \mathcal{P} \to \mathbb{R}$ *is a polynomial*

of degree $\leq \dim(M_{\mathbb{R}}) + k$ in the sense of Definition 5.1 (however, the continuity assumption might not be satisfied in general).

We now apply the results of [32] summarized above to our situation. Note that $I_f: \mathcal{P}^+ \to \mathbb{R}$ can be naturally extended to a homomorphism of additive groups $Z(M_{\mathbb{R}}) \to \mathbb{R}$ via

$$I_f(\alpha) = \int_{M_{\mathbb{R}}} f(x) \cdot \alpha(x) d\mu = \sum_{i=1}^k n_i \int_{P_i} f(x) d\mu \quad \text{for } \alpha = \sum_{i=1}^k n_i \mathbb{1}_{P_i} \in Z(M_{\mathbb{R}}).$$

Furthermore, $I_f: Z(M_{\mathbb{R}}) \to \mathbb{R}$ is a polynomial finitely additive measure of degree $\leq \deg(f)$ which follows by the observation that the following map is a polynomial of degree $\leq \deg(f)$ for any $\alpha = \sum_{i=1}^k n_i \mathbb{1}_{P_i} \in Z(M_{\mathbb{R}})$:

$$I_f(\tau(\cdot,\alpha)): M_{\mathbb{R}} \to \mathbb{R}; \quad \lambda \mapsto \int_{\tau(\lambda,\alpha)} f(x) d\mu = \sum_{i=1}^k n_i \cdot \int_{P_i + \lambda} f(x) d\mu$$
$$= \sum_{i=1}^k n_i \cdot \int_{P_i} f(x - \lambda) d\mu.$$

By Theorem 5.4, the restriction of I_f to the space of virtual polytopes is a polynomial of degree $\leq \dim(M_{\mathbb{R}}) + \deg(f)$. The continuity assumption of Definition 5.1 can be seen as follows.

By [32, Corollary 2.2], the natural inclusion $\mathcal{P}^+ \to Z(M_{\mathbb{R}})$; $P \mapsto \mathbb{1}_P$ extends to an isomorphism of groups from \mathcal{P} (with Minkowski addition) onto the invertible elements \mathcal{P}^* of $Z(M_{\mathbb{R}})$ of degree 1 (equipped with the multiplication operation). The scalar multiplication on \mathcal{P} naturally carries over to \mathcal{P}^*

$$(\mathbb{1}_P)^{\lambda} = \begin{cases} \mathbb{1}_{\lambda P} & \text{if } \lambda \ge 0, \\ \mathbb{1}_{-\lambda P}^{-1} & \text{if } \lambda < 0 \end{cases}$$

for any convex polytope $P \subseteq M_{\mathbb{R}}$ and every real scalar λ .

We obtain an isomorphism of vector spaces between $\mathcal{P} \simeq \mathcal{P}^*$. By our choice of the topology from above, restricting this isomorphism to corresponding finite-dimensional subspaces yields a homeomorphism. Let $U \subseteq \mathcal{P}^*$ be a finite-dimensional subspace and choose a basis $\alpha_1, \ldots, \alpha_n \in U$. Then, two elements $\lambda = \alpha_1^{\lambda_1} \star \cdots \star \alpha_n^{\lambda_n}$ and $\mu = \alpha_1^{\mu_1} \star \cdots \star \alpha_n^{\mu_n}$ in U are close to each other if and only if the scalars λ_i and μ_i are close to each other for any index *i*. It is straightforward to show that $I_f(\lambda)$ and $I_f(\mu)$ are close to each other, i.e., I_f is continuous.

Recall that a polynomial $f: V \to \mathbb{R}$ on a (possibly infinite-dimensional) vector space *V* is called *homogeneous of degree m* if, for all $v \in V$ and every $t \in \mathbb{R}$, we have $f(t \cdot v) = t^m \cdot f(v)$.

To summarize the above, let us state the following theorem.

Theorem 5.5 If $f: M_{\mathbb{R}} \to \mathbb{R}$ is a homogeneous polynomial of degree *m*, then the function $I_f: \mathcal{P}^+ \to \mathbb{R}; \Delta \mapsto I_f(\Delta) = \int_{\Delta} f(x) d\mu$ admits a unique extension to a homogeneous polynomial of degree n + m on \mathcal{P} .

6 Proof of the BKK Theorems

In this section, we prove Theorem 4.1. For the reader's convenience, we recall the used notation. Let $p: E \to B$ be a principal torus bundle with respect to an *n*-dimensional torus *T* over a smooth compact orientable manifold *B* of real dimension *k*. The character lattice of *T* was denoted by *M*. Let $\Sigma \subseteq N_{\mathbb{R}} = \text{Hom}_{\mathbb{Z}}(M, \mathbb{R})$ be a smooth complete fan with rays ρ_1, \ldots, ρ_s . The primitive ray generators we denote by e_1, \ldots, e_s . Let h_1, \ldots, h_s be the basis of \mathcal{P}_{Σ} consisting of piecewise linear functions which vanish on every ray of Σ except one where it evaluates with 1 on the corresponding primitive ray generator. Let $X = X_{\Sigma}$ be the toric variety corresponding to Σ and denote the total space of the associated toric bundle by E_X . To keep notation simple, we use the same notation $p: E_X \to B$ for the projection map of the toric bundle. Fix $i \leq \frac{k}{2}$ and a class $\gamma \in H^{k-2i}(B, \mathbb{R})$. The rays ρ_i correspond to divisors in *X* which give rise to divisors D_i in E_X . For a virtual polytope $\Delta \in \mathcal{P}_{\Sigma}$, we introduced

$$\rho(\Delta) = \sum_{i=1}^{s} h_{\Delta}(e_i)[D_i] \in H^2(E_X, \mathbb{R}).$$

where $[D_i]$ is the class dual to $D_i \subseteq E_X$ and $h_\Delta \colon N_{\mathbb{R}} \to \mathbb{R}$ is the support function of Δ . Further, we introduce

$$F_{\gamma} \colon \mathcal{P}_{\Sigma} \to \mathbb{R}; \quad F_{\gamma}(\Delta) = \rho(\Delta)^{n+i} \cdot p^*(\gamma).$$

We noticed that F_{γ} is a homogeneous polynomial of degree n + i on \mathcal{P}_{Σ} .

Recall that for any character $\lambda \in M$, we have an associated complex line bundle \mathcal{L}_{λ} on *B*. Taking Chern classes and extending by linearity, we obtain a morphism of vector spaces $c: M_{\mathbb{R}} \to H^2(B, \mathbb{R})$. Let $f_{\gamma}: M_{\mathbb{R}} \to \mathbb{R}$ be the function $f_{\gamma}(x) = c(x)^i \cdot \gamma$. We defined a map $I_{\gamma}: \mathcal{P}_{\Sigma} \to \mathbb{R}$ which is explicitly given for $\Delta \in \mathcal{P}_{\Sigma}^+$ by

$$I_{\gamma}(\Delta) = \int_{\Delta} f_{\gamma}(x) \mathrm{d}\mu.$$

In the previous section, we saw how to extend this definition to all of \mathcal{P}_{Σ} . Furthermore, by Theorem 5.5, I_{γ} is a homogeneous polynomial of degree n + i on \mathcal{P}_{Σ} .

We prove Theorem 4.1 by induction on the parameter $0 \le i \le \frac{k}{2}$.

Let us start with the base case i = 0. If i = 0, then $\gamma \in H^k(B, \mathbb{R})$ is a multiple of the class dual to a point. For simplicity, let us assume that this multiple is 1. Therefore, $p^*(\gamma)$ is the class dual to a fibre, which is a toric variety, and thus, $\rho(\Delta)^n \cdot p^*(\gamma)$ coincides with the degree of the divisor in X corresponding to Δ . By the classical BKK theorem, this can be computed by the normalized volume of Δ which equals to $n! \cdot I_{\gamma}(\Delta)$. In other words, for i = 0, Theorem 4.1 reduces to the classical BKK theorem.

As induction hypothesis, suppose that we know Theorem 4.1 for some $i - 1 \ge 0$. The induction step consists of proving that Theorem 4.1 is also true for i. Since both F_{γ} and I_{γ} are homogeneous polynomials of the same degree n + i, in order to show equality between $(n + i)! \cdot I_{\gamma}(\Delta)$ and $i! \cdot F_{\gamma}(\Delta)$, it suffices to show that all their partial derivatives of order n coincide. In other words, it suffices to consider differential monomials $\partial_{i_1}^{k_1} \dots \partial_{i_r}^{k_r}$ where $\partial_i = \partial/\partial_{h_i}$ are the partial derivatives along the coordinate vectors of $\mathcal{P}_{\Sigma} \simeq \mathbb{R}^s$ and $\sum_{i=1}^r k_i = n$. Let us call the number $\sum_{i=1}^r (k_i - 1)$ the *multiplicity of the monomial* $\partial_{i_1}^{k_1} \dots \partial_{i_r}^{k_r}$. In particular, a monomial has multiplicity 0 if and only if it is square free.

The proof of equality of the partial derivatives of order *n* of $(n + i)! \cdot I_{\gamma}(\Delta)$ and $i! \cdot F_{\gamma}(\Delta)$ is by induction on the multiplicity *m* of the applied differential monomial. We will refer to the induction over *i* as the "outer induction" and we are going to call the induction over *m* the "inner induction".

The base case of the inner induction (i.e., the case of square free differential monomials) is treated in the next two subsections. The result of these calculations is summarized in Proposition 6.4.

6.1 Differentiation of I_{γ}

The main result of this subsection is Lemma 6.1 below which computes the partial derivative of I_{γ} with respect to square free monomial. Here, we work with a slightly more general situation than needed for the proof of Theorem 4.1: we assume that Σ is a simplicial fan, and we define the function $I_f : \mathcal{P} \to \mathbb{R}$ via $I_f(\delta) = \int_{\delta} f d\mu$ for any smooth function f on $M_{\mathbb{R}}$. In the proof of Theorem 4.1, we use $f = f_{\gamma}$.

Lemma 6.1 Let $I = \{i_1, \ldots, i_r\} \subseteq \{1, \ldots, s\}$ be a subset and k_1, \ldots, k_r positive integers. If Δ is a polytope in the interior of \mathcal{P}^+_{Σ} and $\rho_{i_1}, \ldots, \rho_{i_r}$ do not span a cone in Σ , then we have

$$\partial_{i_1}^{k_1} \cdots \partial_{i_r}^{k_r} \left(I_f |_{\mathcal{P}_{\Sigma}} \right) (\Delta) = 0.$$

However, if r = n and $\rho_{i_1}, \ldots, \rho_{i_n}$ span a cone in Σ dual to the vertex $A \in \Delta$, we have

$$\partial_I \left(I_f | \mathcal{P}_{\Sigma} \right) (\Delta) = f(A) \cdot |\det(e_{i_1}, \dots, e_{i_n})|.$$

Here det $(e_{i_1}, \ldots, e_{i_n})$ denotes the determinant of the matrix whose *j*th column is given by the vector e_{i_j} . Furthermore, ∂_I denotes the partial derivative $\partial/\partial h_{i_1} \cdots \partial/\partial h_{i_n}$ along the coordinate vectors.

The proof of Lemma 6.1 relies on a folklore result on convex chains. We continue to use the notation from above. Fix real numbers $\{\lambda_i\}_{i \in I}$ and define the virtual polytope $h + \sum_{i \in I} \lambda_i h_i$, where *h* denotes the support function of the polytope Δ . If $\{\lambda_i\}_{i \in I}$ are sufficiently small, the virtual polytope $h + \sum_{i \in I} \lambda_i h_i$ is in the interior of \mathcal{P}_{Σ}^+



Fig. 1 Alternating sum of characteristic functions for two adjacent edges



Fig. 2 Alternating sum of characteristic functions for two disjoint edges

in which case it corresponds to the polytope obtained from Δ by moving the facets corresponding to the rays ρ_{i_j} according to the coefficients λ_{i_j} . For any subset $J \subseteq I$, we introduce the virtual polytope $\Delta(J):=h+\sum_{j\in J}\lambda_jh_j$. Furthermore, we define the convex chain

$$\sigma(\lambda_{i_1},\ldots,\lambda_{i_r}) = \sum_{J \subseteq I} (-1)^{r+|I_-|+|J|} \cdot \mathbb{1}_{\Delta(J)} \text{ where } I_- := \{i \in I : \lambda_i < 0\}.$$

The following folklore result will be needed in the proof of Lemma 6.1.

Proposition 6.2 Let $I = \{i_1, \ldots, i_r\} \subseteq \{1, \ldots, s\}$ be a subset and let $\{\lambda_i\}_{i \in I}$ be sufficiently small real numbers. If any $\lambda_i = 0$ for $i \in I$ or $\rho_{i_1}, \ldots, \rho_{i_r}$ do not span a cone in Σ , then $\sigma(\lambda_{i_1}, \ldots, \lambda_{i_r}) = 0$. If all $\lambda_i \neq 0$ for $i \in I$, and the cardinality of I is $n = \dim(N_{\mathbb{R}})$, and $\rho_{i_1}, \ldots, \rho_{i_n}$ span a cone in Σ dual to the vertex $A \in \Delta$, then

$$\sigma(\lambda_{i_1},\ldots,\lambda_{i_n})=\mathbb{1}_{A+\Pi(\lambda_{i_1},\ldots,\lambda_{i_n})}.$$

Here, $A + \Pi(\lambda_{i_1}, \ldots, \lambda_{i_n})$ denotes the half-open parallelepiped spanned by the vectors $\lambda_{i_1}e_{i_1}, \ldots, \lambda_{i_n}e_{i_n}$ shifted by the vertex A

$$\Pi\left(\lambda_{i_1},\ldots,\lambda_{i_n}\right) = \left\{\sum_{i\in I} a_i\lambda_i e_i: 0 < a_i \le 1 \text{ if } \lambda_i > 0 \text{ and } 0 \le a_i < 1 \text{ if } \lambda_i < 0\right\}.$$

See Figs. 1 and 2 for an illustration of Lemma 6.2 in dimension 2.

Sketch of proof For the reader's convenience, we include a sketch of proof.

It is straightforward to check that $\sigma(\lambda_{i_1}, \ldots, \lambda_{i_r})$ vanishes if any $\lambda_i = 0$ for $i \in I$. Hence, suppose no λ_i vanishes for $i \in I$. It turns out to be more convenient to work with the case that all $\lambda_i > 0$ for $i \in I$. We can always reduce to this case as follows. Let $I_-:=\{i \in I : \lambda_i < 0\}$ and set $\Delta' = h + \sum_{i \in I_-} \lambda_i h_i$. Let h' be the support function of Δ' and define $\Delta'(J) = h' + \sum_{j \in J} |\lambda_j| h_j$ for any subset $J \subseteq I$. Consider the convex chain

$$\sigma'(|\lambda_{i_1}|,\ldots,|\lambda_{i_r}|) = \sum_{J\subseteq I} (-1)^{r+|J|} \mathbb{1}_{\Delta'(J)}.$$

It is straightforward to verify that $\sigma(\lambda_{i_1}, \ldots, \lambda_{i_r}) = (-1)^{|I_-|} \cdot \sigma'(|\lambda_{i_1}|, \ldots, |\lambda_{i_r}|)$, and thus, we may assume from now on that $\lambda_i > 0$ for every $i \in I$.

Suppose that the facet F_i of Δ corresponding to the ray $\rho_i \in \Sigma$ is given by the supporting hyperplane $H_i = \{x \in M_{\mathbb{R}} : \langle x, e_i \rangle = c_i\}$ for some $c_i \in \mathbb{R}$. Note that the polytope $\Delta(I)$ is obtained from Δ by moving the supporting hyperplanes H_i for $i \in I$ outwards. We say that a point $x \in \Delta(I)$ is beyond the hyperplane H_i for $i \in I$ if $\langle x, e_i \rangle > c_i$. We get a map from the points of $\Delta(I)$ to the powerset 2^I of I

$$\Delta(I) \to 2^I; x \mapsto \mathcal{I}(x) := \{i \in I : \langle x, e_i \rangle > c_i\}$$

For $J \subseteq I$, define the region $\mathcal{C}(J)$ as the set of points $x \in \Delta(I)$, where $\mathcal{I}(x) = J$ is constant. The regions $\mathcal{C}(J)$ form a (disjoint) decomposition of $\Delta(I)$ into (possibly half-open) polytopes. Points in these regions behave well with respect to the characteristic functions $\mathbb{1}_{\Delta(J)}$ which appear in the convex chain $\sigma(\lambda_{i_1}, \ldots, \lambda_{i_r})$

$$\mathbb{1}_{\Delta(J)} = \begin{cases} 1 & \text{if } \mathcal{I}(x) \subseteq J, \\ 0 & \text{otherwise} \end{cases} \quad \text{for } x \in \Delta(I) \text{ and } J \subseteq I.$$

For $x \in \Delta(I)$, a straightforward computation yields

$$\sigma(\lambda_{i_1},\ldots,\lambda_{i_r})(x) = \begin{cases} 0 & \text{if } |\mathcal{I}(x)| < r, \\ (-1)^r & \text{if } |\mathcal{I}(x)| = r. \end{cases}$$

Hence, $\sigma(\lambda_{i_1}, \ldots, \lambda_{i_r}) = (-1)^r \cdot \mathbb{1}_{\mathcal{C}(I)}$. It is straightforward to show that $\mathcal{C}(I) = \emptyset$ if $\rho_{i_1}, \ldots, \rho_{i_r}$ do not span a cone in Σ . If $|I| = n = \dim(N_{\mathbb{R}})$ and $\rho_{i_1}, \ldots, \rho_{i_n}$ span a cone in Σ dual to the vertex $A \in \Delta$, then a straightforward computation shows that $\mathcal{C}(I) = A + \Pi(\lambda_{i_1}, \ldots, \lambda_{i_n})$.

Proof of Lemma 6.1 To keep notation simple, assume $\rho_{i_j} = \rho_j$.

For a monomial $\partial_1^{k_1} \cdots \partial_r^{k_r}$, let $\partial_I = \partial_1 \cdots \partial_r$ be the corresponding square-free monomial. It is enough to show that $\partial_I F_{\gamma}(\Delta) = 0$ to prove that $\partial_1^{k_1} \cdots \partial_r^{k_r} F_{\gamma}(\Delta) = 0$.

For a smooth function g on \mathcal{P}_{Σ} , the partial derivative $\partial_I g(x)$ can be expressed as

$$\partial_I g(x) = \lim_{\lambda_1, \dots, \lambda_r \to 0} \left(\frac{1}{\lambda_1 \cdots \lambda_r} D_{\lambda_1 h_1} \dots D_{\lambda_r h_r} g(x) \right), \tag{3}$$

where $D_{\lambda_i h_i}$ denotes the difference operator introduced above and $\lambda_1, \ldots, \lambda_r \in \mathbb{R}$. Applying equality (3) to the function $I_f|_{\mathcal{P}_{\Sigma}}$, one gets

$$\partial_I I_f(\Delta) = \lim_{\lambda_1,\dots,\lambda_r \to 0} \frac{1}{\lambda_1 \cdots \lambda_r} \int_{M_{\mathbb{R}}} f(x) \cdot (-1)^{|I_-|} \cdot \sigma(\lambda_1,\dots,\lambda_r)(x) d\mu,$$

where $\sigma(\lambda_1, \ldots, \lambda_r)$ is the convex chain defined above and $I_- = \{i = 1, \ldots, r : \lambda_i < 0\}$. By Proposition 6.2, the convex chain $\sigma(\lambda_1, \ldots, \lambda_r)$ vanishes if ρ_1, \ldots, ρ_r do not form a cone in Σ . By the same proposition, if r = n and ρ_1, \ldots, ρ_n generate a cone in Σ dual to the vertex $A \in \Delta$, it is equal to the characteristic function $\mathbb{1}_{A+\Pi(\lambda_1,\ldots,\lambda_n)}$ of the half-open parallelepiped spanned by the vectors $\lambda_1 e_1, \ldots, \lambda_n e_n$ at the point A. In this latter case, i.e., ρ_1, \ldots, ρ_n span a cone in Σ , let $T : \mathbb{R}^n \to M_{\mathbb{R}}$ be the diffeomorphism given by $T(x_1, \ldots, x_n) = x_1 e_1 + \cdots + x_n e_n + A$. Then, by a straightforward calculation, we obtain

$$\int_{A+\Pi(\lambda_1,\dots,\lambda_n)} f(x) d\mu$$

= $(-1)^{|I_-|} \cdot \int_0^{\lambda_1} \cdots \int_0^{\lambda_n} f(T(x_1,\dots,x_n)) \cdot |\det(e_1,\dots,e_n)| dx_1 \cdots dx_n,$

where $det(e_1, \ldots, e_n)$ is the determinant of the Jacobian matrix of *T*. Therefore, the lemma follows by the fundamental theorem of calculus:

$$\lim_{\lambda_1,\dots,\lambda_n\to 0} \frac{1}{\lambda_1\dots\lambda_n} \int_{A+\Pi(\lambda_1e_1,\dots,\lambda_ne_n)} (-1)^{|I_-|} \cdot f(x) \mathrm{d}\mu = f(A) \cdot |\det(e_1,\dots,e_n)|.$$

6.2 Differentiation of F_{γ}

Next, we verify the base case of square-free differential monomials for F_{γ} .

Lemma 6.3 Let $I = \{i_1, \ldots, i_r\} \subseteq \{1, \ldots, s\}$ be a subset and k_1, \ldots, k_r positive integers. If Δ is a polytope in the interior of \mathcal{P}_{Σ}^+ , such that $\rho_{i_1}, \ldots, \rho_{i_r}$ do not span a cone in Σ , we have

$$\partial_{i_1}^{k_1} \cdots \partial_{i_r}^{k_r} \left(F_{\gamma} |_{\mathcal{P}_{\Sigma}} \right) (\Delta) = 0.$$

If $r = n = \dim(N_{\mathbb{R}})$ and $\rho_{i_1}, \ldots, \rho_{i_n}$ span a cone in Σ dual to the vertex $A \in \Delta$, we have

$$\partial_I \left(F_{\gamma} |_{\mathcal{P}_{\Sigma}} \right) (\Delta) = \frac{(n+i)!}{i!} f_{\gamma}(A).$$

Proof To keep notation simple, assume $i_j = j$ for j = 1, ..., r.

Deringer

For a monomial $\partial_1^{k_1} \cdots \partial_r^{k_r}$, let $\partial_I = \partial_1 \cdots \partial_r$ be the corresponding square-free monomial. It is enough to show that $\partial_I F_{\gamma}(\Delta) = 0$ to prove that $\partial_1^{k_1} \cdots \partial_r^{k_r} F_{\gamma}(\Delta) = 0$.

We compute the expansion of the polynomial F_{γ} at Δ . This amounts to expressing $F_{\gamma}(\Delta + \sum_{i=1}^{s} \lambda_i h_i)$ in terms of the monomials $\lambda_1^{\alpha_1} \cdots \lambda_s^{\alpha_s}$ for non-negative integers $\alpha_1, \ldots, \alpha_s$. A straightforward computation yields

$$F_{\gamma}\left(\Delta + \sum_{i=1}^{s} \lambda_{i}h_{i}\right) = \left(\rho(\Delta) + \sum_{i=1}^{s} \lambda_{i}[D_{i}]\right)^{n+i} \cdot p^{*}(\gamma)$$

$$= \sum_{\alpha_{0} + \alpha_{1} + \dots + \alpha_{s} = n+i} \binom{n+i}{\alpha_{0}, \alpha_{1}, \dots, \alpha_{s}} \cdot \rho(\Delta)^{\alpha_{0}} \cdot [D_{1}]^{\alpha_{1}} \cdots$$

$$[D_{s}]^{\alpha_{s}} \cdot p^{*}(\gamma) \cdot \lambda_{1}^{\alpha_{i}} \cdots \lambda_{s}^{\alpha_{s}}$$

$$= \frac{(n+i)!}{(n+i-r)!} \cdot \rho(\Delta)^{n+i-r} \cdot [D_{1}] \cdots [D_{r}] \cdot p^{*}(\gamma) \cdot \lambda_{1} \cdots \lambda_{r}$$

$$+ (\text{other terms}),$$

where $\binom{n+i}{\alpha_0,\alpha_1,\ldots,\alpha_s} = \frac{(n+i)!}{\alpha_0!\cdot\alpha_1!\cdot\cdot\cdot\alpha_s!}$ denotes the usual multinomial coefficient. The derivative $\partial_I F_{\gamma}(\Delta)$ is equal to the coefficient in front of the monomial $\lambda_1 \cdots \lambda_r$ in the expression $F_{\gamma}(\Delta + \sum_{i=1}^{s} \lambda_i h_i)$:

$$\partial_I F_{\gamma}(\Delta) = \frac{(n+i)!}{(n+i-r)!} \cdot \rho(\Delta)^{n+i-r} \cdot [D_1] \cdots [D_r] \cdot p^*(\gamma).$$

Since Σ is a smooth fan, the divisors D_1, \ldots, D_r intersect transversely in E_X . Therefore, the product $[D_1] \cdots [D_r]$ is the class Poincaré dual to the set theoretic intersection of these divisors. In the case that e_1, \ldots, e_r do not generate a cone in Σ the set theoretic intersection of D_1, \ldots, D_r is empty, and so, $\partial_I F_{\gamma}(\Delta) = 0$.

For the proof of the second part, let $r = n = \dim(N_{\mathbb{R}})$. We have

$$\partial_I F_{\gamma}(\Delta) = \frac{(n+i)!}{i!} \cdot \rho(\Delta)^i \cdot [D_1] \cdots [D_n] \cdot p^*(\gamma).$$

If e_1, \ldots, e_n generate a cone in Σ dual to the vertex A of Δ , then $[D_1] \cdots [D_n] = E_A$, where $E_A = E \times_T A$ is the torus invariant submanifold in E_X corresponding to A. In particular, the restriction of the projection map $p: E_A \to B$ is a diffeomorphism.

Let $h_{\widetilde{\Delta}}$ be the support function of the polytope $\widetilde{\Delta} = \Delta - A$ which is the translation of the polytope Δ for which the vertex A is at the origin. Since the vertex of $\widetilde{\Delta}$ corresponding to A is at the origin, we get

$$h_{\widetilde{\Delta}}(e_1) = \ldots = h_{\widetilde{\Delta}}(e_n) = 0$$
, and so $\rho(\widetilde{\Delta}) = \sum_{j>n} h_{\widetilde{\Delta}}(e_j) \cdot [D_j].$

Deringer

Hence, by the first part, we get $\rho(\widetilde{\Delta}) \cdot [D_1] \cdots [D_n] = 0$ as there is no cone in Σ with more than *n*-rays. Therefore

$$\rho(\Delta)^{i} \cdot [D_{1}] \cdots [D_{n}] \cdot p^{*}(\gamma) = \rho(\widetilde{\Delta} + A)^{i} \cdot [D_{1}] \cdots [D_{n}] \cdot p^{*}(\gamma)$$
$$= \rho(A)^{i} \cdot E_{A} \cdot p^{*}(\gamma).$$

By Proposition 3.5, $\rho(A) = p^* c(A)$. Since $p: E_A \to B$ is a diffeomorphism, we get

$$\rho(A)^{l} \cdot p^{*}(\gamma) \cdot E_{A} = (p^{*}c(A))^{l} \cdot p^{*}(\gamma) \cdot E_{A} = c(A)^{l} \cdot \gamma = f_{\gamma}(A),$$

and therefore, $\partial_I F_{\gamma}(\Delta) = \frac{(n+i)!}{i!} f_{\gamma}(A).$

The base case of the inner induction is an immediate corollary of Lemmas 6.1 and 6.3:

Corollary 6.4 (Base case of the inner induction) For any $i \leq \frac{k}{2}$, any $\gamma \in H^{k-2i}(B, \mathbb{R})$ and any square-free differential monomial $\partial_I = \partial_{i_1} \dots \partial_{i_n}$ of order n (where $I = \{i_1, \dots, i_n\} \subseteq \{1, \dots, s\}$), we have

$$\partial_I \left((n+i)! \cdot I_{\gamma}(\Delta) \right) = \partial_I \left(i! \cdot F_{\gamma}(\Delta) \right).$$

6.3 The Inner Induction Step

In this subsection, the index set $I \subseteq \{1, \ldots, s\}$ is considered a multiset. As induction hypothesis, suppose that $\partial_I ((n+i)! \cdot I_{\gamma}(\Delta)) = \partial_I (i! \cdot F_{\gamma}(\Delta))$ for any differential monomial ∂_I of multiplicity $m-1 \ge 0$. It remains to show that the equality is true for differential monomials of multiplicity m. As before, to keep notation simple, assume $i_j = j$ for $j = 1, \ldots, r$, so that $\partial_I = \partial_1^{k_1} \ldots \partial_r^{k_r}$ for positive integers k_1, \ldots, k_r . By reordering the coordinates of \mathcal{P}_{Σ} , we may assume $k_1 > 1$. By Lemmas 6.1 and 6.3, it suffices to consider the case where the vectors e_1, \ldots, e_r form a cone in Σ (as otherwise $\partial_I I_{\gamma}(\Delta) = 0 = \partial_I F_{\gamma}(\Delta)$).

The plan is to express ∂_1 in terms of a Lie derivative L_v (for some $v \in M_{\mathbb{R}}$) and other partial derivatives. Then, the (inner) induction step will follow by an explicitly computation of $L_v I_{\gamma}(\Delta)$ and $L_v F_{\gamma}(\Delta)$.

As e_1, \ldots, e_r form a cone in the smooth fan Σ , they can be completed to a basis of $N_{\mathbb{R}}$. We take v to be the first vector of the dual basis of $M_{\mathbb{R}}$. Then, $\langle v, e_1 \rangle = 1$, and $\langle v, e_j \rangle = 0$ for $j = 2, \ldots, r$. Since $v \in M_{\mathbb{R}}$ (considered as an element of \mathcal{P}_{Σ}) is given by $v = \sum_{i=1}^{s} \langle v, e_i \rangle h_i$, we obtain that $L_v = \sum_{i=1}^{s} \langle v, e_i \rangle \partial_i$, and thus, $\partial_1 = L_v - \sum_{j>r} \langle v, e_j \rangle \partial_j$. We get

$$\partial_{I} = \partial_{1}^{k_{1}} \dots \partial_{r}^{k_{r}} = \left(L_{v} - \sum_{j > r} \langle v, e_{j} \rangle \partial_{j} \right) \partial_{1}^{k_{1}-1} \cdot \partial_{2}^{k_{2}} \dots \partial_{r}^{k_{r}}$$
$$= L_{v} \cdot \partial_{1}^{k_{1}-1} \cdot \partial_{2}^{k_{2}} \dots \partial_{r}^{k_{r}} - \sum_{j > r} \langle v, e_{j} \rangle \cdot \partial_{1}^{k_{1}-1} \cdot \partial_{2}^{k_{2}} \dots \partial_{r}^{k_{r}} \cdot \partial_{j}.$$

Springer

Since $k_1 > 1$ and j > r, each monomial in the sum $\sum_{j>r} \langle v, e_j \rangle \partial_1^{k_1-1} \cdot \partial_2^{k_2} \cdots \partial_r^{k_r} \cdot \partial_j$ has multiplicity m - 1, so by the induction hypothesis (of the inner induction), we get

$$(n+i)! \cdot \left(\sum_{j>r} \langle v, e_j \rangle \partial_1^{k_1-1} \cdot \partial_2^{k_2} \cdots \partial_r^{k_r} \cdot \partial_j \right) \cdot I_{\gamma}(\Delta)$$

= $i! \cdot \left(\sum_{j>r} \langle v, e_j \rangle \partial_1^{k_1-1} \cdot \partial_2^{k_2} \cdots \partial_r^{k_r} \cdot \partial_j \right) \cdot F_{\gamma}(\Delta).$

It remains to consider the first summand

Lemma 6.5 In the situation above, we have $(n + i)! \cdot L_v I_{\gamma}(\Delta) = i! \cdot L_v F_{\gamma}(\Delta)$.

Proof. A direct calculation shows

$$\begin{split} L_{v}I_{\gamma}(\Delta) &= \partial_{t} \Big|_{t=0} \left(\int_{\Delta+tv} c(x)^{i} \cdot \gamma \, \mathrm{d}\mu \right) = \partial_{t} \Big|_{t=0} \left(\int_{\Delta} c(x+tv)^{i} \cdot \gamma \, \mathrm{d}\mu \right) = \\ &= \partial_{t} \Big|_{t=0} \left(\int_{\Delta} \left(\sum_{a=0}^{i} {i \choose a} t^{a} \cdot c(v)^{a} c(x)^{i-a} \right) \cdot \gamma \, \mathrm{d}\mu \right) \\ &= \int_{\Delta} i \cdot c(v) c(x)^{i-1} \cdot \gamma \, \mathrm{d}\mu \\ &= i \cdot \int_{\Delta} c(x)^{i-1} \cdot (c(v) \cdot \gamma) \, \mathrm{d}\mu = i \cdot I_{c(v) \cdot \gamma}(\Delta). \end{split}$$

Similarly, by a direct calculation and Proposition 3.5, we get

$$\begin{split} L_{v}F_{\gamma}(\Delta) &= \partial_{t}\Big|_{t=0} \Big(\rho(\Delta+tv)^{n+i} \cdot p^{*}(\gamma)\Big) = \partial_{t}\Big|_{t=0} \Big((\rho(\Delta)+t\rho(v))^{n+i} \cdot p^{*}(\gamma)\Big) = \\ &= \partial_{t}\Big|_{t=0} \left(\left(\sum_{a=0}^{n+i} \binom{n+i}{a} t^{a} \cdot \rho(v)^{a} \cdot \rho(\Delta)^{n+i-a}\right) \cdot p^{*}(\gamma)\right) \\ &= (n+i) \cdot \rho(v) \cdot \rho(\Delta)^{n+i-1} \cdot p^{*}(\gamma) \\ &= (n+i) \cdot \rho(\Delta)^{n+i-1} \cdot p^{*}(c(v) \cdot \gamma) = (n+i) \cdot F_{c(v)\gamma}(\Delta). \end{split}$$

Since $c(v) \cdot \gamma \in H^{k-2(i-1)}(B, \mathbb{R})$, by the induction hypothesis of the outer induction, we have

$$(n+i-1)! \cdot I_{c(v)\cdot\gamma}(\Delta) = (i-1)! \cdot F_{c(v)\cdot\gamma}(\Delta).$$

Therefore

$$(n+i)! \cdot L_{v}I_{\gamma}(\Delta) = (n+i)! \cdot i \cdot I_{c(v)\cdot\gamma}(\Delta) = i! \cdot (n+i) \cdot F_{c(v)\cdot\gamma}(\Delta) = i! \cdot L_{v}F_{\gamma}(\Delta). \quad \Box$$

7 Graded-Commutative n-Self-Dual Algebras

7.1 Introduction

Suppose A is a cohomology ring of a compact oriented manifold which can be represented as factor-algebra of some other algebra B. In this section, we discuss the question what extra information is needed to determine A out of B? For the reader's convenience, we recall several crucial notions.

Definition 7.1 A graded \mathbb{K} -algebra $A = A^0 \oplus A^1 \oplus \cdots \oplus A^k \oplus \cdots$ over a field \mathbb{K} is called *graded-commutative* if, for any homogeneous elements x, y, the following relation hold:

$$xy = (-1)^{\deg(x) + \deg(y)} yx,$$

where deg(x) (resp. deg(y)) denotes the degree of x (resp. of y).

Such algebras are fundamental in geometry. For example, the cohomology ring $A = H^*(X, \mathbb{K})$ of a CW complex X is known to be a graded-commutative \mathbb{K} -algebra. However, the cohomology ring of compact oriented manifolds satisfies further important properties:

Definition 7.2 Let *n* be a natural number. A graded-commutative \mathbb{K} -algebra is *Poincaré n*-self-dual (or just, is *n*-self-dual) if the following conditions hold:

- (1) The algebra A has a multiplicative unit element e which is homogeneous of degree zero, i.e., $e \in A^0 \subseteq A$.
- (2) The homogeneous components A^k for k > n vanish, i.e., $A^k = 0$ for k > n, and $\dim_{\mathbb{K}}(A^n) = 1$.
- (3) The pairing $A^k \times A^{n-k} \to A^n$ induced by multiplication in the algebra A is non-degenerate for $0 \le k \le n$.

Remark 7.3 By the non-degeneracy of the pairing $A^k \times A^{n-k} \to A^n$, we have $\dim_{\mathbb{K}}(A^k) = \dim_{\mathbb{K}}(A^{n-k})$ for every $0 \le k \le n$, such that the component A^k is a finite-dimensional space. In particular, $\dim_{\mathbb{K}}(A^0) = 1$.

The following example plays a key role in our discussion.

Example 7.4 Let M be a connected compact oriented n-dimensional manifold. By Poincaré duality, the cohomology ring $A = H^*(M, \mathbb{K})$ is an n-self-dual graded-commutative \mathbb{K} -algebra. Moreover, A is equipped with two extra structures:

- A K-linear function ℓ_{*}: A → K given as follows. By linearity, it suffices to define ℓ_{*} on homogeneous elements. For homogeneous α of degree n, we let ℓ_{*}(α) be equal to the value of the cohomology class α on the fundamental class of the oriented manifold M. If the degree of α is not n, we set ℓ_{*}(α) = 0.
- (2) A K-bilinear intersection form F_{ℓ*} on A defined by the identity F_{ℓ*}(α, β) = ℓ_{*}(α ⋅ β) (recall that we denote the cup product of the cohomology ring A by "·"). Note that F_{ℓ*} is non-degenerate (the non-degeneracy of F_{ℓ*} is equivalent to the Poincaré duality on A).

Consider another graded-commutative \mathbb{K} -algebra $B = B^0 \oplus B^1 \oplus \cdots \oplus B^k \oplus \cdots$ with multiplicative unit element $e \in B^0$ and $\dim_{\mathbb{K}}(B^0) = 1$. Suppose that the cohomology ring $A = H^*(M, \mathbb{K})$ from Example 7.4 is a factor-algebra of B.

Question 7.5 We are going to investigate the following questions:

- (1) What extra information on B is needed to determine the algebra A?
- (2) What extra information on *B* is needed to determine the intersection form F_{ℓ^*} on *A*?

We provide answers to both questions in Sects. 7.2 and 7.3, respectively.

7.2 Graded-Commutative n-Self-Dual Factor-Algebras

In this section, let $B = B^0 \oplus B^1 \oplus \cdots \oplus B^k \oplus \cdots$ be a graded-commutative K-algebra with multiplicative unit element $e \in B^0$ and $\dim_{\mathbb{K}}(B^0) = 1$. We want to describe all *n*-self dual factor-algebras of *B*. To that extent, we introduce the following notion.

Definition 7.6 An ideal $I \subseteq B$ is called *n*-self-dual (or *n*-sd ideal for short) if I is a two-sided homogeneous ideal and the factor-algebra A = B/I is *n*-self-dual.

Clearly, factor-algebras of B which are n-self-dual are in correspondence with n-sd ideals. Therefore, part (1) of Question 7.5 boils down to describing all n-sd ideals in B. Our solution consists of the following steps:

- (1) If *I* is an *n*-sd ideal, then $I \cap B^n$ is a hyperplane in B^n (see Lemma 7.7).
- (2) Conversely, for any hyperplane $L \subseteq B^n$, there is a unique *n*-sd ideal I = I(L), such that $I \cap B^n = L$ (see Lemmas 7.12 and 7.13).
- (3) We present an explicit construction of the *n*-sd ideal I(L) (see Definition 7.11).

Lemma 7.7 If $I \subseteq B$ is an n-sd ideal, then $L = I \cap B^n$ is a hyperplane in B^n .

Proof Since $B^n/(I \cap B^n) = A^n$ and dim $(A^n) = 1$, L is a hyperplane in B^n . \Box

As hyperplanes are in correspondence with linear functions, we introduce the following:

Definition 7.8 A linear function $\ell: B \to \mathbb{K}$ on a graded-commutative \mathbb{K} -algebra B is *n*-homogeneous if it is not identically equal to zero on B, but for any $m \neq n$, its restriction to the homogeneous component B^m vanishes. Denote by $L_{\ell} \subseteq B^n$ the hyperplane in B^n defined by identity $L_{\ell} = \{\ell = 0\} \cap B^n$.

Clearly, any hyperplane $L \subseteq B^n$ is equal to the hyperplane L_{ℓ} for a unique (up to a non-zero scalar multiple) *n*-homogeneous linear function ℓ . Next, we explain how to obtain an *n*-self-dual factor-algebra from an *n*-homogeneous linear function $\ell : B \to \mathbb{K}$.

Definition 7.9 For an *n*-homogeneous linear function $\ell: B \to \mathbb{K}$, let $I_1(L_\ell)$ and $I_2(L_\ell)$ be the subsets of *B* defined by the following conditions:

- (1) an element $a \in B$ belongs to $I_1(L_\ell)$ if and only if $\ell(ab) = 0$ for all $b \in B$;
- (2) an element $b \in B$ belongs to $I_2(L_\ell)$ if and only if $\ell(ab) = 0$ for all $a \in B$.

Clearly, the sets $I_1(L_\ell)$ and $I_2(L_\ell)$ depend only on the hyperplane $L = L_\ell \subseteq B^n$.

Lemma 7.10 For an *n*-homogeneous linear function ℓ , $I_1(L_\ell) = I_2(L_\ell)$ is a two-sided homogeneous ideal in *B*.

Proof We first prove that the set $I_1(L_\ell)$ is a *right* ideal in *B*. If $a \in I_1(L_\ell)$ and $b \in B$, then for any $c \in B$, the identity $\ell((ab)c) = \ell(a(bc)) = 0$ holds, and thus, $ab \in I_1(L_\ell)$. If $a, b \in I_1(L_\ell)$, then $\ell(ac) = \ell(bc) = 0$ for any $c \in B$. Hence, $\ell((a + b)c) = 0$ which implies $a + b \in I_1(L_\ell)$.

Next, we prove that $I_1(L_\ell)$ is homogeneous, i.e., if $a = \sum_{i=0}^d a_i$ for $a_i \in B^i$ is in $I_1(L_\ell)$, then $a_i \in I_1(L_\ell)$ for every *i*. For $c = \sum_{m=0}^{d'} c_m$ with $c_m \in B^m$, we have $\ell(a_ic) = \sum_{m=0}^{d'} \ell(a_ic_m)$, and thus, it suffices to show $\ell(a_ic_m) = 0$ for any homogeneous $c_m \in B_m$. Since ℓ is *n*-homogeneous, $\ell(a_ic_m) = 0$ for every *i* with $i + m \neq n$. However, from $0 = \ell(ac_m) = \sum_{i=0}^d \ell(a_ic_m)$, it follows $\ell(a_{n-m}c_m) = 0$. Thus, $a_i \in I_1(L_\ell)$.

Analogously, it follows that $I_2(L_\ell)$ is a homogeneous *left* ideal in *B*.

Let us show that $I_1(L_\ell) \subseteq I_2(L_\ell)$. Let $a = \sum_{j=0}^d a_j$ be in $I_1(L_\ell)$ and $b = \sum_{i=0}^{d'} b_i$ be in *B*, where $a_i, b_i \in B^i$. Then, $\ell(ba) = \sum_{i=0}^{d'} \sum_{j=0}^d \ell(b_i a_j) = \sum_{i=0}^{d'} \sum_{j=0}^d (-1)^{i+j} \ell(a_j b_i)$. Since $I_1(L_\ell)$ is homogeneous, we have $a_j \in I_1(L_\ell)$, and thus, $\ell(a_j b_i) = 0$ for all couples *i*, *j*. It follows $\ell(ba) = 0$, which means that $I_1(L_\ell) \subseteq I_2(L_\ell)$.

Similarly, we can prove that $I_2(L_\ell) \subseteq I_1(L_\ell)$ and the statement is proven. \Box

Definition 7.11 For an *n*-homogeneous linear function ℓ , let $I(L_{\ell})$ be the two-sided ideal $I_1(L_{\ell}) = I_2(L_{\ell})$. The ideal $I(L_{\ell})$ depends only on the hyperplane $L = L_{\ell}$, and so, the notation $I(L) = I(L_{\ell})$ makes sense.

Lemma 7.12 For any hyperplane $L \subseteq B^n$, the set $I(L) \subseteq B$ is an n-sd ideal.

Proof By Lemma 7.10, I(L) is a two-sided homogeneous ideal. Set A = B/I(L).

If k > n, then for any $b_k \in B^k$ and any $c = \sum_{i=0}^d c_i$ in B with $c_i \in B^i$, we have $\ell(b_k c) = \sum_{i=0}^d \ell(b_k c_i) = 0$ (since ℓ is *n*-homogeneous). Hence, $B^k = I(L) \cap B^k$ for k > n, or equivalently A^k vanishes for k > n.

It remains to show that the pairing $A^k \times A^{n-k} \to A^n$ induced by multiplication is non-degenerate. Suppose $a \in A^k$ is in the kernel of this pairing, i.e., ab = 0 for any $b \in A^{n-k}$. Let $a^* \in B^k$ be an element whose image in A is equal to a. Then, for any $b^* \in B^{n-k}$, we have $a^*b^* \in I(L)$, and in particular, $\ell(a^*b^*) = 0$. Since ℓ is *n*-homogeneous, for any $c \in B$, we have $\ell(a^*c) = \ell(a^*c_{n-k})$, where $c_{n-k} \in B^{n-k}$ is the homogeneous component of c of degree n - k. Thus, $a^* \in I_1(L_\ell) = I(L)$, or equivalently a = 0 in A, i.e., the pairing is non-degenerate in the first component. \Box

Lemma 7.13 For any hyperplane $L \subseteq B^n$, there is a unique n-sd ideal $I \subseteq B$, such that $I \cap B^n = L$. This ideal I coincides with the ideal I(L).

Proof Let ℓ be an *n*-homogeneous linear function, such that $L = L_{\ell}$.

We start by showing that $I(L) \cap B^n = L$. The inclusion " \subseteq " is straightforward. For the reverse inclusion, suppose $a \in L$. Then, for any $b = \sum_{i=0}^{d} b_i$ in B with $b_i \in B^i$, we have $\ell(ab) = \sum_{i=0}^{d} \ell(ab_i) = \ell(ab_0) = b_0 \ell(a) = 0$, and thus, $a \in I(L) \cap B^n$. This shows the existence-part of the statement.

To show uniqueness, let $I \subseteq B$ be an *n*-sd ideal with $I \cap B^n = L$. As both I and I(L) are homogeneous, it suffices to show that $I \cap B^k = I(L) \cap B^k$ for every k. If $a \in I(L) \cap B^k = I_1(L_\ell) \cap B^k$, then its image in the factor-algebra A = B/I belongs to the kernel of the pairing $A^k \times A^{n-k} \to A^n = B/L$. Thus, $a \in I \cap B^k$ and $I_1(L_\ell) \cap B^k \subseteq I \cap B^k$. Conversely, if $a \notin I_1(L_\ell) \cap B^k$, then there is $b \in B^{n-k}$, such that $ab \notin L$. Thus, the image of the element $ab \in B^n$ is not equal to zero in A = B/I and $a \notin I \cap B^k$.

7.3 Frobenius Forms and *n*-Self-Dual Factor-Algebras

In this section, we investigate the second part of Question 7.5. For the reader's convenience, we provide a summary at the end of the section.

Consider a bilinear form $F: L_1 \times L_2 \to \mathbb{K}$ on a product of \mathbb{K} -linear spaces L_1 and L_2 . Recall the *left radical* $R_1 \subseteq L_1$, respectively, the *right radical* $R_2 \subseteq L_2$ of F

(1) $a \in R_1$ if and only if F(a, b) = 0 for all $b \in L_2$;

(2) $b \in R_2$ if and only if F(a, b) = 0 for all $a \in L_1$.

Definition 7.14 Let $\ell: B \to \mathbb{K}$ be an *n*-homogeneous linear function on a gradedcommutative algebra *B*. The *Frobenius bilinear form* $F_{\ell}: B \times B \to \mathbb{K}$ associated with ℓ is the form defined by the identity $F_{\ell}(a, b) = \ell(a \cdot b)$.

Theorem 7.15 Let $\ell: B \to \mathbb{K}$ be an *n*-homogeneous linear function on a gradedcommutative algebra *B*.

- (1) The n-sd ideal $I(L_{\ell}) = I_1(L_{\ell}) = I_2(L_{\ell})$ coincides with the left and also the right radical of F_{ℓ} .
- (2) There is a unique n-homogeneous linear function ℓ_* on the factor-algebra A = B/I(L), such that $\ell = \rho^* \ell_*$, where $\rho: B \to A$ is the natural homomorphism. Furthermore, for any $a, b \in B$, the relation $F_{\ell}(a, b) = F_{\ell_*}(\rho(a), \rho(b))$ holds.
- (3) The Frobenius form F_{ℓ_*} on A is non-degenerate.

Proof By definition, the left radical (resp. the right radical) of the form F_{ℓ} coincides with $I_1(L_{\ell})$ (resp. with $I_2(L_{\ell})$). By Lemma 7.10, $I(L_{\ell}) = I_2(L_{\ell}) = I_2(L_{\ell})$, and thus, the first statement follows (see also Definition 7.11).

For the second statement, observe that the kernel $L \subseteq B^n$ of the restriction of the function $\ell|_{B^n}$ coincides with the kernel of the surjective map $\rho: B^n \to A^n$ (see Lemma 7.13). Thus, there is a unique function $\ell_*: A^n \to \mathbb{K}$, such that $\ell = \rho^* \ell_*$. The relation $F_{\ell}(a, b) = F_{\ell^*}(\rho(a), \rho(b))$ follows by definition.

For the last statement, observe that the form F_{ℓ_*} is obtained from the form F_{ℓ} by taking the quotient of *B* by the left and right radical of the form F_{ℓ} . Thus, F_{ℓ_*} is a non-degenerate form.

Let us summarize the results from this section. To reconstruct an *n*-sd factor-algebra A = B/I equipped with an intersection form on *A*, it suffices to fix a Frobenius form F_{ℓ} on *B* corresponding to an *n*-homogeneous linear function ℓ on *B*. The algebra *A* is equal to $B/I(L_{\ell})$ (note that the algebra *A* depends only on the hyperplane $L = L_{\ell}$, i.e., proportional *n*-homogeneous functions define the same algebra *A*). The form F_{ℓ} is induced by the unique non-degenerate intersection form F_{ℓ_*} on *A*. The set of *n*-sd factor-algebras A = B/I equipped with non-degenerate intersection forms is in one-to-one correspondence with the set of *n*-homogeneous linear functions on *B*.

Theorem 7.16 Let B be a graded-commutative algebra over a field \mathbb{K} , and let A be an n-sd factor-algebra of B with a chosen isomorphism $\phi: A^n \to \mathbb{K}$. Let $\rho: B \to A$ be the natural homomorphism. Then

$$A \simeq B/I(L_\ell),$$

where ℓ is the n-homogeneous linear function defined by

$$\ell \colon B^n \to \mathbb{K}, \ \ell(b) = \phi(\rho(b)),$$

and extended by 0 to B^i with $i \neq n$.

Moreover, for any n-homogeneous linear function ℓ , the algebra $A = B/I(L_{\ell})$ is n-self dual and the corresponding Frobenius form F_{ℓ_*} on A is non-degenerate.

8 Applications

8.1 The Cohomology Ring of Toric Bundles

In this section, we apply the results from Sects. 6 and 7 to give a description of the cohomology ring of a toric bundle. Let, as before, $p: E \to B$ be a principal torus bundle, Σ a smooth projective fan with $\Sigma(1) = \{\rho_1, \ldots, \rho_r\}, X = X_{\Sigma}$ the corresponding toric variety, and E_X the corresponding toric bundle. By the Leray–Hirsch theorem (see Theorem 3.1), the cohomology ring $H^*(E_X, \mathbb{R})$ is a quotient of the polynomial algebra $R[x_1, \ldots, x_r]$, where $R = H^*(B, \mathbb{R})$ is the cohomology ring of the base.

Theorem 8.1 In the notation from above, the cohomology ring of E_X is given by

$$H^*(E_X,\mathbb{R})\simeq R[x_1,\ldots,x_r]/I(L_\ell),$$

where $\ell : R[x_1, \ldots, x_r] \to \mathbb{R}$ is a (k + 2n)-homogeneous linear function defined by

$$\ell(\gamma \cdot x_{i_1} \cdots x_{i_s}) = I_{\gamma}(\rho_{i_1}, \dots, \rho_{i_s})$$

for any monomial $\gamma \cdot x_{i_1} \cdots x_{i_s}$ with $\deg(\gamma) + 2s = k + 2n$.

Proof By Theorem 7.16, as $H^*(E_X, \mathbb{R})$ is a graded commutative (k + 2n)-self dual factor-algebra of $R[x_1, \ldots, x_r]$, it is given by $R[x_1, \ldots, x_r]/I(L_\ell)$ for a (k + 2n)-homogeneous linear function ℓ which is obtained by pairing cohomology classes with the fundamental class of E_X . Hence, the statement follows by Theorem 4.1.

We conclude this subsection with a proof of Theorem 3.8, i.e., a proof of the Sankaran-Uma-description of the cohomology ring of toric bundles.

Proof of Theorem 3.8 As above, let *R* denote the cohomology ring of the base. We want to show that the cohomology ring $H^*(E_X, \mathbb{R})$ is isomorphic to the quotient $R[x_1, \ldots, x_r]/(I + J)$, where

$$I = \langle x_{j_1} \cdots x_{j_k} : \rho_{j_1}, \dots, \rho_{j_k} \text{ do not span a cone of } \Sigma \rangle$$
$$J = \left\langle -c \left(\lambda\right) + \sum_{i=1}^r \langle e_i, \lambda \rangle x_i : \lambda \in M \right\rangle.$$

By the Leray–Hirsch theorem (Theorem 3.1), there is a surjection $\phi \colon R[x_1, \ldots, x_r] \to H^*(E_X, \mathbb{R})$. By Proposition 3.5, the ideal *J* is contained in ker(ϕ). Furthermore, by Theorem 8.1, the polarization formula (1), and the first part of Lemma 6.1, we have $I \subseteq \text{ker}(\phi)$. Thus, we have a surjection

$$\phi \colon R[x_1, \ldots, x_r]/(I+J) \to H^*(E_X, \mathbb{R}).$$

To simplify the notation, we also denote this surjection with ϕ . It remains to show that the surjection ϕ is an isomorphism, i.e., ϕ is injective. For that it suffices to verify that $\dim_{\mathbb{R}} R[x_1, \ldots, x_r]/(I + J) = \dim_{\mathbb{R}} H^*(E_X, \mathbb{R}).$

Let $SR_{\Sigma} := \mathbb{R}[x_1, \dots, x_r]/I$ be the Stanley–Reisner algebra associated to Σ . Then

$$A:=R[x_1,\ldots,x_r]/(I+J)=(R\otimes SR_{\Sigma})/J.$$

We consider the ring $A_0 := R \otimes (SR_{\Sigma}/J_0)$ where J_0 is the ideal given by $J_0 = \langle \sum_{i=1}^r \langle e_i, \lambda \rangle x_i : \lambda \in M \rangle$. By the Künneth formula, we have that

$$A_0 = R \otimes (SR_{\Sigma}/J_0) = R \otimes H^*(X, \mathbb{R}) \simeq H^*(B \times X, \mathbb{R}).$$

By the Leray–Hirsch theorem (Theorem 3.1), we have $\dim_{\mathbb{R}}(R \otimes H^*(X, \mathbb{R})) = \dim_{\mathbb{R}} H^*(E_X, \mathbb{R})$, and thus

$$\dim_{\mathbb{R}} A_0 = \dim_{\mathbb{R}} H^*(B \times X, \mathbb{R}) = \dim_{\mathbb{R}} H^*(E_X, \mathbb{R}) \le \dim_{\mathbb{R}} A.$$

Hence, it suffices to show that $\dim_{\mathbb{R}} A_0 \ge \dim_{\mathbb{R}} A$. We have $A = (R \otimes SR_{\Sigma})/J$ and $A_0 = (R \otimes SR_{\Sigma})/J_0$, with J and J_0 homogeneous ideals. Therefore, to show that $\dim_{\mathbb{R}} A_0 \ge \dim_{\mathbb{R}} A$, it suffices to verify

$$\dim_{\mathbb{R}} J^i \geq \dim_{\mathbb{R}} J_0^i,$$

where $J^i = (R \otimes SR_{\Sigma})^i \cap J$, $J_0^i = (R \otimes SR_{\Sigma})^i \cap J_0$ are the *i*th homogeneous components of J and J_0 , respectively.

Denote by $s_{\lambda} = -c(\lambda) + \sum_{i=1}^{r} \langle e_i, \lambda \rangle x_i$ and $s_{\lambda}^0 = \sum_{i=1}^{r} \langle e_i, \lambda \rangle x_i$ the generators of the ideals J and J_0 , respectively. Note that $R \otimes SR_{\Sigma}$ is a bi-graded ring and $s_{\lambda} = s_{\lambda}^0 - c(\lambda)$ is the difference of two bi-homogeneous elements with deg $(c(\lambda)) =$ $(2, 0), \text{deg}(s_{\lambda}^0) = (0, 2)$. Suppose $v_1, \ldots, v_N \in J_0$ is an \mathbb{R} -basis of J_0^i . For all k = $1, \ldots, N$, we may assume v_k is bi-homogeneous of degree $(n_k, i - n_k)$ for some $n_k = 0, 1, \ldots, i$. Hence

$$v_{k} = \sum_{\lambda \in M} r_{\lambda,k}^{(n_{k},i-n_{k}-2)} s_{\lambda}^{0} \text{ for bi-homogeneous } r_{\lambda,k}^{(n_{k},i-n_{k}-2)}$$

 $\in R \otimes SR_{\Sigma} \text{ of degree } (n_{k}, i-n_{k}-2).$

Set $w_k := \sum_{\lambda \in M} r_{\lambda,k}^{(n_k,i-n_k-2)} s_\lambda \in J^i$ for k = 1, ..., N and note $w_k = v_k + w'_k$ for bi-homogeneous $w'_k \in R \otimes SR_{\Sigma}$ of degree $(n_k + 2, i - n_k - 2)$. We claim $w_1, ..., w_N$ are \mathbb{R} -linearly independent concluding the proof.

Suppose $(a_1, \ldots, a_N) \neq (0, \ldots, 0) \in \mathbb{R}^N$, such that $\sum_{i=1}^N a_i w_i = 0 \in J^i$. Set $n := \min\{n_k : k = 1, \ldots, N, a_k \neq 0\}$. Then, the (n, i - n)-homogeneous component of $\sum_{i=1}^N a_i w_i$ is given by $\sum_{k \in \mathcal{K}} a_k v_k = 0$ for some subset $\mathcal{K} \subseteq \{1, \ldots, N\}$. A contradiction (as v_1, \ldots, v_N form a basis of J_0^i).

8.2 Ring of Conditions

The ring of conditions constitutes an intersection ring for horospherical homogeneous spaces. We refer to [13] for further details and references. The exposition given here follows [20].

Let *G* be a reductive complex algebraic group and $H \subseteq G$ be a closed subgroup. A cycle of codimension *k* of *G*/*H* is a formal linear combination $k_1X_1 + \cdots + k_rX_r$ of irreducible closed subvarieties X_i of *G*/*H* with $k_i \in \mathbb{Z}$. We write $\mathbb{Z}^k(G/H)$ for the \mathbb{Z} -module of cycles of codimension *k*. Two cycles *X*, *Y* are said to *intersect properly* if either their intersection is empty or the dimension of each irreducible component of $X \cap Y$ equals dim $(X) + \dim(Y) - \dim(G/H)$. We need the following result by Kleiman [25, Corollary 4]:

Theorem 8.2 (Kleiman's transversality theorem) For any two irreducible subvarieties $X, Y \subseteq G/H$, there exists a dense open $U \subseteq G$, such that $g \cdot X$ and Y intersect properly for each $g \in U$. If X, Y have complementary dimensions, $g \cdot X \cap Y$ is a finite union of points and $|g \cdot X \cap Y|$ is constant for generic $g \in G$.

Theorem 8.2 gives rise to an intersection pairing between algebraic cycles of complementary codimensions

$$\mathcal{Z}^{k}(G/H) \times \mathcal{Z}^{\dim(G/H)-k}(G/H) \to \mathbb{Z}; \quad (X, Y) \mapsto (X \cdot Y) := |g \cdot X \cap Y|$$

(for generic $g \in G$),

where $X, Y \subseteq G/H$ are assumed to be irreducible subvarieties.

Two algebraic cycles $X, Y \in \mathbb{Z}^k(G/H)$ are said to be equivalent if, for any algebraic cycle of complementary codimension $Z \in \mathbb{Z}^{\dim(G/H)-k}(G/H)$, the intersection products coincide, i.e., $(X \cdot Z) = (Y \cdot Z)$. We denote the group of equivalence classes by $C^k(G/H)$ ("the group of conditions of codimension k"). The intersection pairing factors through the equivalence relation, so that we obtain a pairing: $C^k(G/H) \times C^{\dim(G/H)-k}(G/H) \to \mathbb{Z}$.

De Concini and Procesi showed that $C^*(G/H) := \bigoplus_{k=0}^{\dim(G/H)} C^k(G/H)$ can be equipped with a ring structure:

Theorem 8.3 [13, Section 6.3] Suppose G/H is a horospherical homogeneous space. Define an intersection product on $C^*(G/H)$: $[X] \cdot [Y] := [gX \cap Y]$ for generic $g \in G$, where $X, Y \subseteq G/H$ are irreducible subvarieties. This intersection product is well defined and there is a canonical isomorphism of graded rings

$$C^*(G/H) = \varinjlim_E H^*(E, \mathbb{Z}),$$

where the limit is taken over complete (or equivalently smooth) horospherical embeddings $G/H \hookrightarrow E$.

Theorem 8.3 is a special case of [13, Section 6.3] which suffices for our purposes. Furthermore, the intersection product in Theorem 8.3 might not be well defined for arbitrary homogeneous spaces G/H. We refer to [20, Exercise 4.1] for a counterexample. In this paper, we will exclusively work with the ring of conditions with real coefficients. Therefore, we introduce the notation $C^*_{\mathbb{R}}(G/H)$ for the tensor product $C^*(G/H) \otimes_{\mathbb{Z}} \mathbb{R}$.

For the rest of this section, let $H \subseteq G$ be a horospherical subgroup. In Sect. 3.2, we saw that $P = N_G(H)$ is a parabolic subgroup and that E = G/H is a principal torus bundle over G/P with respect to the torus T = P/H. For a *T*-toric variety *X*, we saw that the associated toric bundle E_X is a horospherical variety. Since any toric variety is dominated by a smooth projective one, the same is true for these horospherical varieties E_X . Furthermore, using the Luna–Vust theory of spherical embeddings (see, for instance, [28, Theorem 3.3]), it follows that an (arbitrary) horospherical variety $G/H \hookrightarrow Y$ is dominated by some horospherical variety E_X as described above. Thus, the smooth projective E_X forms a cofinal set for the direct limit in Theorem 8.3.

We now give a description of the ring of conditions $C^*(G/H)$ using Theorem 8.1. Let $R = H^*(G/P, \mathbb{Z})$ be the cohomology ring of the flag variety G/P and Sym^{*}(\mathcal{P}) be the symmetric algebra of integral virtual polytopes. Furthermore set $k = \dim_{\mathbb{C}}(G/P)$ and $n = \dim_{\mathbb{C}}(P/H)$.

Corollary 8.4 With the notation above, $C^*(G/H)$ is an 2(n+k)-self dual factor algebra of $R \otimes \text{Sym}(\mathcal{P})$

 $C^*(G/H) = R \otimes \operatorname{Sym}(\mathcal{P})/I_\ell$, with $\ell(\gamma \otimes \Delta_{i_1} \cdots \Delta_{i_s}) = I_{\gamma}(\Delta_{i_1}, \dots, \Delta_{i_s}).$

In particular, $\ell(\gamma \otimes \Delta_{i_1} \cdots \Delta_{i_s}) = 0$ unless deg $(\gamma) + 2s = 2(n+k)$.

Remark 8.5 Note that, in Theorem 8.1, we compute the cohomology ring of toric bundles over \mathbb{R} . Since spherical varieties come equipped with a \mathbb{C}^* -action that has only finitely many fixed points, by the Białynicki–Birula decomposition [2], we have a paving by affine spaces. Thus, $H^*(E_X, \mathbb{Z})$ is torsion free in this case, and hence, our calculations also work over \mathbb{Z} .

We conclude this section with an alternative description of $C^*_{\mathbb{R}}(G/H)$ using the description of the cohomology ring from Sect. 3.2. The challenge is to find a good description of the ring in Theorem 3.8 as we want to take the direct limit over all smooth projective E_X . The following approach is inspired by [8] (see also [20]).

As in Sect. 3.2, let M be the character lattice of the torus T = P/H and set $M_{\mathbb{R}} = M \otimes \mathbb{R}$ and $N_{\mathbb{R}} = \text{Hom}_{\mathbb{Z}}(M, \mathbb{R})$. Let Σ be a smooth projective fan in $N_{\mathbb{R}}$. A map $f: N_{\mathbb{R}} \to \mathbb{R}$ is *piecewise polynomial* if, for any $\sigma \in \Sigma$, the map $f|_{\sigma}: \sigma \to \mathbb{R}$ extends to a polynomial function on the linear space span_{$\mathbb{R}}{\sigma}$, i.e., a piecewise polynomial function f on Σ is a collection of compatible polynomial functions $f_{\sigma}: \sigma \to \mathbb{R}$. In particular, such a function is continuous. We denote by PP_{Σ} the set of all piecewise polynomial functions on Σ which is a ring under pointwise addition and multiplication. Let Sym($M_{\mathbb{R}}$) be the symmetric algebra of polynomial functions on $N_{\mathbb{R}}$. Note that PP_{Σ} is a positively graded \mathbb{R} -algebra with graded subalgebra Sym($M_{\mathbb{R}}$). Indeed, any piecewise polynomial functions.</sub>

We can now reformulate Theorem 3.8:

Proposition 8.6 [20, Proposition 4.10] If E_X is a horospherical variety obtained as associated toric bundle of a smooth projective toric variety $X = X_{\Sigma}$, the cohomology ring $H^*(E_X, \mathbb{R})$ is isomorphic as an $H^*(G/P, \mathbb{R})$ -algebra to the quotient of $H^*(G/P, \mathbb{R}) \otimes PP_{\Sigma}$ by the ideal $\langle c(\lambda) \otimes 1 - 1 \otimes \langle \cdot, \lambda \rangle : \lambda \in M \rangle$.

Let PP be the set of all piecewise polynomial functions on smooth projective fans in $N_{\mathbb{R}}$, i.e., PP = $\bigcup_{\Sigma} PP_{\Sigma}$ where the union is taken over all smooth projective fans $\Sigma \subseteq N_{\mathbb{R}}$.

Theorem 8.7 [20, Theorem 4.11] We have that

$$C^*_{\mathbb{R}}(G/H) \simeq \left(H^*(G/P, \mathbb{R}) \otimes \operatorname{PP} \right) / \langle c(\lambda) \otimes 1 - 1 \otimes \langle \cdot, \lambda \rangle \colon \lambda \in M \rangle,$$

where $\langle \cdot, m \rangle \in \text{Sym}(M_{\mathbb{R}})$ is a globally linear function on $N_{\mathbb{R}}$.

Remark 8.8 Such a description of the ring of conditions as in Theorem 8.7 was certainly known to De Concini and Procesi, although we could not find any references. Their approach relied on a computation of equivariant cohomology of toroidal compactifications of horospherical homogeneous spaces.

9 Cohomology Rings of Toric Bundles with H^{*}(B, ℝ) Generated in Degree 2

The classical BKK theorem is concerned with intersections of divisors in toric varieties: it computes the self-intersection polynomial on $H^2(X_{\Sigma})$, i.e., $f([D]) = [D]^{\dim X_{\Sigma}}$

for divisors $D \subseteq X_{\Sigma}$. Here, we use our generalization of BKK to compute the selfintersection polynomial on the second cohomology group of a toric bundle. We then use this to compute the cohomology ring of toric bundles for which $H^*(B, \mathbb{R})$ is generated in degree 2.

9.1 Self-Intersection Polynomial on $H^2(E_X)$

The Leray–Hirsch theorem (Theorem 3.1) yields a surjection

$$\bar{\rho} \colon H^2(B,\mathbb{R}) \oplus \mathcal{P}_{\Sigma} \to H^2(E_X,\mathbb{R}), \quad (\gamma,\Delta) \mapsto p^*(\gamma) + \rho(\Delta).$$

Therefore, it suffices to compute the self-intersection polynomial $f: H^2(B, \mathbb{R}) \oplus \mathcal{P}_{\Sigma} \to \mathbb{R}$ given by f(x) = 0 if *B* has odd (real) dimension and $f(x) = \bar{\rho}(x)^{n+s}$ if the (real) dimension of *B* is even, say k = 2s. Clearly, the case of even dimensions $\dim_{\mathbb{R}} B = k = 2s$ needs further consideration.

Let $W \subseteq V$ be two \mathbb{R} -vector spaces, and $\Sigma \subseteq W^*$ be a fan. Let us denote by $\mathcal{AP}^+_{W,\Sigma}$ the *cone of W*-affine polytopes, i.e., the cone of polytopes $\Delta \subseteq V$ whose affine span is parallel to W, such that the normal fan of a translation of Δ into W is a coarsening of Σ . Let us denote by $\mathcal{AP}^+_{W,\Sigma}$ the linear subspace of the vector space of virtual polytopes in V generated by $\mathcal{AP}^+_{W,\Sigma}$. Similarly, \mathcal{AP}^+_W (resp. \mathcal{AP}_W) denotes the cone of all polytopes (resp. the subspace of all virtual polytopes) in V whose affine span is parallel to W. Every virtual polytope $\Delta \in \mathcal{AP}_W$ admits a (non-unique) representation $\Delta = \Delta_0 + v$ for some virtual polytope Δ_0 in W and $v \in V$.

Remark 9.1 Although the above representation of Δ as sum $\Delta_0 + v$ is not unique, the image of v under the natural projection $\pi: V \to V/W$ is well defined. In particular, a choice of a complementary subspace $W^{\perp} \subseteq V$ to W induces a decomposition of vector spaces

$$\mathcal{AP}_{W,\Sigma} = \mathcal{P}_{W,\Sigma} \oplus W^{\perp}$$
 and $\mathcal{AP}_{W} = \mathcal{P}_{W} \oplus W^{\perp}$.

Let us fix a Lebesgue measure μ on W. We can regard μ as a translation invariant Lebesgue measure on any affine subspace in V parallel to W. Thus, any smooth function $g: V \to \mathbb{R}$ gives rise to an integral functional

$$I_g: \mathcal{AP}_W \to \mathbb{R}, \quad \Delta \mapsto \int_\Delta g \mathrm{d}\mu.$$

In particular, if $g: V \to \mathbb{R}$ is a homogeneous polynomial of degree d, then, by Theorem 5.5 and Remark 9.1, I_g is a homogeneous polynomial on \mathcal{AP}_W of degree $d + \dim(W)$.

Let $p: E \to B$ be a torus bundle and let Σ be a smooth projective fan with associated toric variety X and corresponding toric bundle E_X . Consider a pair of vector spaces

$$W = M_{\mathbb{R}} \subseteq M_{\mathbb{R}} \oplus H^2(B, \mathbb{R}) = V.$$

As before, we have a linear map $\bar{\rho} \colon \mathcal{AP}_{M_{\mathbb{R}},\Sigma} \to H^2(E_X,\mathbb{R})$ given by

$$\bar{\rho}(\Delta) = \rho(\Delta_0) + p^*(\gamma),$$

where $\Delta = \Delta_0 + \gamma$ for a virtual polytope Δ_0 in $M_{\mathbb{R}}$ and $\gamma \in H^2(B, \mathbb{R})$. Note, by Remark 9.1, the representation of Δ as sum $\Delta_0 + \gamma$ is unique, and hence, the map $\bar{\rho}$ is well defined. Furthermore, we have a map $\bar{c} \colon V \to H^2(B, \mathbb{R})$

$$\bar{c}(\lambda, \gamma) = c(\lambda) + \gamma.$$

Theorem 9.2 In the notation as above, we have

$$s! \cdot \bar{\rho}(\Delta)^{n+s} = (n+s)! \cdot \int_{\Delta} \bar{c}(x)^s \mathrm{d}\mu$$

for any $\Delta \in \mathcal{AP}_{M_{\mathbb{R}},\Sigma}$.

Proof. By Remark 9.1, we can write $\Delta = \Delta_0 + \gamma$ for unique $\Delta_0 \in \mathcal{P}_{M_{\mathbb{R}},\Sigma}$ and $\gamma \in H^2(B, \mathbb{R})$. Thus, we have

$$\bar{\rho}(\Delta)^{n+s} = (\rho(\Delta_0) + p^*(\gamma))^{n+s} = \sum_{i=0}^{s} \binom{n+s}{i} \underbrace{p^*(\gamma)^i \cdot \rho(\Delta_0)^{n+s-i}}_{F_{\gamma^i}(\Delta_0)}$$
$$= \frac{(n+s)!}{s!} \cdot \sum_{i=0}^{s} \binom{s}{i} I_{\gamma^i}(\Delta_0),$$

where in the last equality follows by Theorem 4.1. On the other hand, we have

$$\begin{split} \int_{\Delta} \bar{c}(x)^{s} \mathrm{d}\mu &= \int_{\Delta_{0}+\gamma} \bar{c}(x)^{s} \mathrm{d}\mu = \int_{\Delta_{0}} \bar{c}(x+\gamma)^{s} \mathrm{d}\mu \\ &= \int_{\Delta_{0}} (c(x)+\gamma)^{s} \mathrm{d}\mu = \sum_{i=0}^{s} \binom{s}{i} I_{\gamma^{i}}(\Delta_{0}). \end{split}$$

Therefore, we obtain

$$s! \cdot \bar{\rho}(\Delta)^{n+s} = (n+s)! \cdot \sum_{i=0}^{s} {s \choose i} I_{\gamma^i}(\Delta_0) = (n+s)! \cdot \int_{\Delta} \bar{c}(x)^s \mathrm{d}\mu. \quad \Box$$

9.2 Cohomology Rings of Toric Bundles with H*(B) Generated in Degree 2

Let us discuss a description of the cohomology ring of toric bundles when $H^*(B)$ is generated in degree 2. Then, all odd degrees of $H^*(B, \mathbb{R})$ vanish, and thus, the (real) dimension of *B* is even, say k = 2s. Furthermore, the cohomology ring $H^*(E_X, \mathbb{R})$ of the toric bundle E_X associated with a toric variety $X = X_{\Sigma}$ for a smooth projective fan Σ is also generated in degree 2 (use, e.g., the Leray–Hirsch Theorem, see Theorem 3.1). If the complex dimension of the torus is *n*, then the real dimension of E_X is even, namely 2(n + s).

Since all odd degrees of $H^*(E_X, \mathbb{R})$ vanish, we consider the grading $H^{*/2}(E_X, \mathbb{R})$, i.e., we consider degrees divided by 2. Note that, in this situation, a ring is graded-commutative if and only if the ring is commutative. These notational changes will allow more ergonomic statements below.

Our goal is to extend the description of the cohomology ring of smooth projective toric varieties in terms of differential operators and the volume polynomial to this situation (see Theorem 2.1). For the reader's convenience, we briefly recall the crucial steps of this description.

Let *V* be a finite-dimensional vector space over a field \mathbb{K} of characteristic 0 and let Diff(*V*) be the ring of differential operators on *V* with constant coefficients. A choice of a basis for *V* induces an isomorphism of Diff(*V*) with the ring of polynomials in symbols ∂_i corresponding to partial derivatives along the *i*th coordinate (induced by the chosen basis). The ring Diff(*V*) acts naturally on the ring Sym(*V**) of polynomials on *V*. To describe the cohomology ring of the toric bundle E_X in the case that *c* is surjective, we use the following theorem which is an analog of Theorem 7.16 for commutative algebras which are generated in degree 1. It is implicitly contained in the work of Pukhlikov and the second author (see [33], see also [21, Theorem 1.1.]).

Theorem 9.3 Let A be a commutative finite-dimensional graded algebra over \mathbb{K} . Write $A = \bigoplus_{i=0}^{n} A^{i}$ where A^{i} is the *i*th graded piece of A. Suppose the following conditions hold:

- (1) A is generated (as an algebra) by A^1 , and $A^0 \simeq A^n \simeq \mathbb{K}$.
- (2) The bilinear map $A^i \times A^{n-i} \to A^n \simeq \mathbb{K}$ given by $(u, v) \mapsto u \cdot v$ is non-degenerate for all i = 0, ..., n.

If $\rho: V \to A^1$ is a surjective linear map, then A is isomorphic to Diff(V)/Ann(f)as graded algebra. Here, $f(v):=\rho(v)^n \in A^n \simeq \mathbb{K}$ is a homogeneous polynomial of degree n on V and $\text{Ann}(f) = \{d \in \text{Diff}(V): df = 0\}$ is its annihilator.

We provide a proof to illustrate how Theorem 9.3 follows from Theorem 7.16.

Proof Since the algebra A is generated in degree 1, the surjective linear map $\rho: V \to A^1$ induces a surjection $p: \text{Diff}(V) \simeq \text{Sym}(V) \to A$ of graded algebras. By Theorem 7.16, there is an isomorphism $A \simeq \text{Diff}(V)/I(L_\ell)$ where $\ell: \text{Diff}(V) \to \mathbb{K}$ is uniquely defined by its values on the *n*th graded piece $\text{Diff}^n(V)$ of Diff(V), namely $\ell = \phi \circ p: \text{Diff}^n(V) \to \mathbb{K}$ ($\phi: A^n \to \mathbb{K}$ a fixed isomorphism). It remains to show $I(L_\ell) = \text{Ann}(f)$. Since both ideals are homogeneous, it suffices to check $I(I_\ell) \cap \text{Diff}^j(V) = \text{Ann}(f) \cap \text{Diff}^j(V)$ for all j.

Let $v_1, \ldots, v_n \in V$ and recall L_{v_i} denotes the Lie derivative with respect to the vector v_i (see Definition 4.2). The value $\ell(L_{v_1} \cdots L_{v_n})$ at the corresponding differential monomial is given by polarization of the polynomial $f: V \to \mathbb{K}, v \mapsto \phi(\rho(v)^n)$ (notice the implicit choice of an isomorphism $\phi: A^n \to \mathbb{K}$ in the statement)

$$\ell(L_{v_1}\cdots L_{v_n})=\frac{1}{n!}L_{v_1}\cdots L_{v_n}f.$$

Extending by linearity from the product of linear forms to all of $\text{Diff}^n(V)$, we obtain

$$\ell(\mathbf{d}) = \frac{1}{n!} \mathbf{d}f$$
 for any $\mathbf{d} \in \text{Diff}^n(V)$.

Clearly, for any j > n, we have $I(L_{\ell}) \cap \text{Diff}^{j}(V) = \text{Ann}(f) \cap \text{Diff}^{j}(V) = \text{Diff}^{j}(V)$.

Let $d \in \operatorname{Ann}(f)$ be a differential operator of degree $j \leq n$. Then, d'df = 0for any $d' \in \operatorname{Diff}(V)$, and thus, $d \in I(L_{\ell})$. If $d \notin \operatorname{Ann}(f)$, then df is a non-zero polynomial of degree n - j. There exists $d' \in \operatorname{Diff}^{n-j}(V)$ with $d'df \neq 0$. It follows that $\ell(d'd) \neq 0$ which implies that $d \notin I(L_{\ell})$. Therefore, $I(L_{\ell}) = \operatorname{Ann}(f)$ and the theorem is proved.

Theorems 9.2 and 9.3 yield the following description of the cohomology ring $H^*(E_X, \mathbb{R})$ of toric bundles over a base manifold with cohomology ring generated in degree 2.

Corollary 9.4 Let $p: E \to B$ be a torus bundle, Σ be a smooth projective fan, and E_X be the associated toric bundle. Suppose $H^2(B, \mathbb{R})$ is generated in degree 2. Then, the cohomology ring $H^*(E_X, \mathbb{R})$ is given by

$$H^{*/2}(E_X,\mathbb{R})\simeq \operatorname{Diff}(\mathcal{AP}_{M_{\mathbb{R}},\Sigma})/\operatorname{Ann}(I),$$

where $I: \mathcal{AP}_{M_{\mathbb{R}},\Sigma} \to \mathbb{R}$ is the homogeneous polynomial of degree n + s given by $I(\Delta) = \int_{\Delta} \bar{c}(x)^{s} d\mu$.

Proof By Remark 9.1, we have $\mathcal{AP}_{M_{\mathbb{R}},\Sigma} = \mathcal{P}_{M_{\mathbb{R}},\Sigma} \oplus H^2(B,\mathbb{R})$, and thus, the map $\bar{\rho}: \mathcal{AP}_{M_{\mathbb{R}},\Sigma} \to H^2(E_X,\mathbb{R})$ is surjective. Moreover, by Theorem 9.2, we have

$$s! \cdot \bar{\rho}(\Delta)^{n+s} = (n+s)! \cdot \int_{\Delta} \bar{c}(x)^s \mathrm{d}\mu,$$

so the statement follows from Theorem 9.3.

Remark 9.5 From Proposition 3.5, it straightforwardly follows that in Corollary 9.4 we may replace $V = M_{\mathbb{R}} \oplus H^2(B, \mathbb{R})$ by $V' = M_{\mathbb{R}} \oplus c(M_{\mathbb{R}})^{\perp}$ where $c(M_{\mathbb{R}})^{\perp} \subseteq$ $H^2(B, \mathbb{R})$ is a complementary subspace of $c(M_{\mathbb{R}})$ in $H^2(B, \mathbb{R})$. Then, $c(M_{\mathbb{R}})^{\perp} \cong$ $(H^2(B, \mathbb{R})/c(M_{\mathbb{R}}))$. In particular, if $c: M_{\mathbb{R}} \to H^2(B, \mathbb{R})$ is surjective, the cohomology ring $H^*(E_X, \mathbb{R})$ is given by

$$H^{*/2}(E_X,\mathbb{R})\simeq \operatorname{Diff}(\mathcal{P}_{M_{\mathbb{R}},\Sigma})/\operatorname{Ann}(I),$$

where $I: \mathcal{P}_{M_{\mathbb{R}},\Sigma} \to \mathbb{R}$ is the homogeneous polynomial of degree n + s given by $I(\Delta) = \int_{\Delta} c(x)^s d\mu$.

10 Examples

Here, we consider several concrete applications of our results. Section 9 will enable us to retrieve a version of the Brion–Kazarnovskii theorem [7, 22] for horospherical varieties. For the special case of toric bundles over a full flag variety, we obtain two descriptions of the cohomology ring: one via integrals of the Weyl polynomial, and another one relying on the theory of string polytopes. The section is concluded by a reproduction of the famous description of the cohomology ring of projective bundles.

10.1 The Brion-Kazarnovskii Theorem for Horospherical Varieties

Let *G* be a connected semisimple simply connected group, *T* be a maximal torus, and *B* be a Borel subgroup containing *T*. Let $\Phi \subseteq M(T)$ be the corresponding set of roots and $S \subseteq \Phi$ be the set of simple roots corresponding to our choice of Borel *B*. Recall that parabolic subgroups containing *B* are in one-to-one correspondence with subsets of simple roots. Finally, let C^+ be the positive Weyl chamber in $M_{\mathbb{R}}(T)$.

Fix a parabolic subgroup $P_I \subseteq G$ containing *B* given by the set of simple roots $I \subseteq S$. The inclusion $i = i_I : B \rightarrow P_I$ gives rise to an inclusion of character lattices

$$i^*: M(P_I) \to M(B) \simeq M(T),$$

where the image is the sublattice of M(T) given by (here, $\check{\alpha}$ denotes the coroot associated to the simple root α)

$$i^*(M(P_I)) = \bigcap_{\alpha \in I} \ker \check{\alpha} \subseteq M(T).$$

From now on, we will identify the character lattice $M(P_I)$ with its image $i^*(M(P_I))$ in M(T). It is well known that the horospherical homogeneous spaces that are torus principal bundles over G/P_I are classified by sublattices $\Lambda \subseteq M(P_I)$ in the character lattice of P_I . As with $M(P_I)$, we will identify Λ with the corresponding sublattice of M(T). For a given sublattice $\Lambda \subseteq M(P_I)$, the horospherical subgroup $H_{\Lambda} \subseteq G$ is defined by

$$H_{\Lambda} = \{ g \in P \mid \lambda(g) = 1 \text{ for all } \lambda \in \Lambda \}.$$

It is easy to see that the normalizer satisfies $N_G(H_\Lambda) = P_I$ and the factor group $T_\Lambda = P_I/H_\Lambda$ is an algebraic torus. Hence, the homogeneous space G/H_Λ has a structure of a T_Λ -principal bundle over G/P_I . We identify the character lattice of T_Λ with Λ via the pull-back

 $\pi^*: M(T_\Lambda) \to M(P_I)$, where $\pi: P_I \to T_\Lambda$ is the natural projection.

Let X be the T_{Λ} -toric variety associated with a fan Σ in Λ and E_X be the corresponding toric bundle. Equivalently, E_X is the toroidal compactification of G/H_{Λ}

corresponding to Σ (regarded as a colored fan). The section concludes with a computation of the self-intersection polynomial on $H^2(E_X, \mathbb{R})$ if Σ is a smooth projective fan.

Let $c_{\Lambda} \colon M_{\mathbb{R}}(T_{\Lambda}) \to H^2(G/P_I, \mathbb{R})$ be as before, i.e., the linear map of vector spaces induced by the homomorphism of latices

$$c_{\Lambda} \colon M(T_{\Lambda}) \to H^2(G/P_I, \mathbb{Z}), \quad \lambda \mapsto c_1(\mathcal{L}_{\lambda}).$$

Note there is a map $c: M_{\mathbb{R}}(P_I) \to H^2(G/P_I, \mathbb{R})$ which associates to a character $\lambda \in M_{\mathbb{R}}(P_I)$ the first Chern class $c_1(\mathcal{L}_{\lambda}) \in H^2(G/P_I, \mathbb{R})$ of the associated line bundle $\mathcal{L}_{\lambda} = G \times_{P_I} \mathbb{C}_{\lambda}$. For any $\Lambda \subseteq M(P_I)$, the linear map $c_{\Lambda}: M_{\mathbb{R}}(T_{\Lambda}) \simeq \Lambda_{\mathbb{R}} \to H^2(G/P_I, \mathbb{R})$ coincides with the restriction of the linear map $c: M_{\mathbb{R}}(P_I) \to H^2(G/P_I, \mathbb{R})$ as $G/H_{\Lambda} \times_{T_{\Lambda}} \mathbb{C}_{\lambda} = G \times_{P_I} \mathbb{C}_{\lambda}$ for any $\lambda \in \Lambda$. We will denote all these maps by the same letter c.

We recall the following classical result by Borel [6]. We found the expository paper [4] useful in our study of this beautiful subject. Let $W = N_G(T)/T$ be the Weyl group of *G* with respect to *T*.

Theorem 10.1 The map $c: M_{\mathbb{R}}(P_I) \to H^2(G/P_I, \mathbb{R})$ is an isomorphism.

Proof By [4, Proposition 1.3] the map $c: M_{\mathbb{R}}(B) \to H^2(G/B, \mathbb{R})$ is an isomorphism. Indeed, by [4, Proposition 1.3], $c: \text{Sym}(M_{\mathbb{R}}(B))/J \to H^{*/2}(G/B, \mathbb{R})$ is an isomorphism of graded algebras where J is the ideal generated by *W*-invariant elements in $\text{Sym}(M_{\mathbb{R}}(B))$ of positive degree. However, there are no non-zero *W*-invariant elements in $M_{\mathbb{R}}(B) \subseteq \text{Sym}(M_{\mathbb{R}}(B))$, and thus, $c: M_{\mathbb{R}}(B) \to H^2(G/B, \mathbb{R})$ is a linear isomorphism.

Let W_I be the subgroup of the Weyl group W generated by the reflections s_{α} for $\alpha \in I$. By [4, Theorem 5.5], the restriction of $c: M_{\mathbb{R}}(B) \to H^2(G/B, \mathbb{R})$ to $M_{\mathbb{R}}(B)^{W_I}$ induces an isomorphism

$$c: M_{\mathbb{R}}(B)^{W_I} \xrightarrow{\simeq} H^2(G/P_I, \mathbb{R}).$$

From above, recall that

$$M(P_I) = \bigcap_{\alpha \in I} \ker \check{\alpha} = \{\lambda \in M(B) \mid \langle \lambda, \check{\alpha} \rangle = 0 \text{ for all } \alpha \in I \}.$$

In other words, the sublattice $M(P_I)$ of M(B) is obtained by intersecting M(B) with the intersection of the reflection hyperplanes of the reflections s_{α} for $\alpha \in I$. This shows $M_{\mathbb{R}}(B)^{W_I} = M_{\mathbb{R}}(P_I)$ as desired.

For a dominant weight $\lambda \in C^+$, let us denote by V_{λ} the irreducible representation of *G* with highest weight λ . We will also need the following well-known result.

Theorem 10.2 [Borel–Weil–Bott theorem, see, e.g., [17, Theorem 6.1]] We have

 $H^0(G/P_I, \mathcal{L}_{\lambda}) = V_{\lambda}$ for any dominant weight $\lambda \in C^+ \cap M(P_I)$.

Weyl showed (see, for instance, [36, Satz 5]) that the map $\lambda \mapsto \dim(V_{\lambda})$ is the restriction of an explicit (inhomogeneous) polynomial on $M_{\mathbb{R}}(T)$ to the set of dominant weights. Let $f_{W,I}$ be the top homogeneous part of the restriction of the Weyl polynomial to $M_{\mathbb{R}}(P_I) \subseteq M_{\mathbb{R}}(T)$. We will need the following classical result.

Proposition 10.3 Let G, P_I and $f_{W,I}$ be as before. Then, for any $\lambda \in M_{\mathbb{R}}(P_I)$, we have

$$c(\lambda)^s = s! \cdot f_{W,I}(\lambda), \text{ where } s = \dim_{\mathbb{C}}(G/P_I).$$

Proof Since both sides of the equality are polynomials, it suffices to verify it on a subset of dominant weights whose convex hull is an affine full-dimensional cone in $M_{\mathbb{R}}(P_I)$. It is well known that the line bundle \mathcal{L}_{λ} associated with $\lambda \in M(P_I)$ is very ample if and only if $\langle \lambda, \check{\alpha} \rangle > 0$ for all $\alpha \in S \setminus I$, i.e., the interior lattice points in the cone $C^+ \cap M_{\mathbb{R}}(P_I)$ yield very ample line bundles. It suffices to verify the above equality on this discrete set of points.

Fix $\lambda \in C^+ \cap M_{\mathbb{R}}(P_I)$ in the interior. The very ample line bundle \mathcal{L}_{λ} gives rise to an embedding of G/P_I into the projective space $\mathbb{P}(V_{\lambda}^*)$ (see Theorem 10.2). Note that the degree of \mathcal{L}_{λ} , i.e., $c(\lambda)^s$, coincides with the degree of the projective variety $G/P_I \hookrightarrow \mathbb{P}(V_{\lambda}^*)$. Let $H_{\lambda}(t)$ be the Hilbert polynomial of the section ring $R_{\lambda} = \bigoplus_{k\geq 0} H^0(G/P_I, \mathcal{L}_{\lambda}^{\otimes k})$. Then, the degree of the projective variety $G/P_I \hookrightarrow \mathbb{P}(V_{\lambda}^*)$ coincides with the s! multiple of the leading coefficient of $H_{\lambda}(t)$. By Theorem 10.2, $R_{\lambda} = \bigoplus_{k\geq 0} V_{k\lambda}$, and thus, $H_{\lambda}(t)$ coincides with the Weyl polynomial, say f_W , restricted to the ray $\mathbb{R}_{\geq 0} \cdot \lambda$, i.e., $f_W(t\lambda) = H_{\lambda}(t)$. The restriction $f_W|_{M_{\mathbb{R}}(P_I)}$ accepts the decomposition into homogeneous components

 $f_W|_{M_{\mathbb{R}}(P_I)} = f_{W,I}$ + homogeneous components of lower degree.

In particular

$$f_W(t\lambda) = f_W|_{M_{\mathbb{D}}(P_I)}(t\lambda) = f_{W,I}(\lambda)t^{\dim(G/P_I)}$$
 + lower terms.

Note $f_{W,I}(\lambda) \neq 0$ as $f_{W,I}$ is of top degree (i.e., of degree dim (G/P_I)). The statement follows.

We are now ready to formulate a version of the Brion–Kasarnovskii theorem [7, 22] for our setting. Let us consider a pair of vector spaces $\Lambda_{\mathbb{R}} \subseteq M_{\mathbb{R}}(P_I)$. By Theorem 10.1, we can identify $M_{\mathbb{R}}(P_I)$ with $H^2(G/P_I, \mathbb{R})$. Let $\Lambda_{\mathbb{R}}^{\perp}$ be a complementary subspace of $\Lambda_{\mathbb{R}}$ in $M_{\mathbb{R}}(P_I)$. By Remark 9.5, we have the surjection

$$\bar{\rho}\colon \mathcal{AP}_{\Lambda_{\mathbb{R}},\Sigma} \simeq \mathcal{P}_{\Lambda_{\mathbb{R}},\Sigma} \oplus \Lambda_{\mathbb{R}}^{\perp} \to H^2(E_X,\mathbb{R}).$$

Hence, $H^*(E_X, \mathbb{R}) = \text{Diff}(\mathcal{AP}_{\Lambda_{\mathbb{R}},\Sigma})/\text{Ann}(f)$, where $f: \mathcal{AP}_{\Lambda,\Sigma} \to H^{2(n+s)}(E_X, \mathbb{R}) \simeq \mathbb{R}$ is the self-intersection polynomial given by $f(\Delta) = \bar{\rho}(\Delta)^{n+s}$ (see Theorem 9.3). It remains to compute f.

Theorem 10.4 In the notation as above, for any $\Delta \in \mathcal{AP}_{\Lambda_{\mathbb{R}},\Sigma}$, we have

$$\bar{\rho}(\Delta)^{n+s} = (n+s)! \cdot \int_{\Delta} f_{W,I}(\lambda) \mathrm{d}\mu,$$

where $n = \operatorname{rk}(\Lambda)$ and μ is Lebesgue measure on $\Lambda_{\mathbb{R}}$ normalized with respect to Λ .

Proof The theorem follows immediately from Theorem 9.2 and Proposition 10.3. \Box

10.2 Horospherical Varieties over Full Flag Variety

As before, let *G* be a connected semisimple simply connected algebraic group, *B* a Borel subgroup in *G*, and $T \subseteq B$ a maximal torus. In the proof of Theorem 10.1, we already used Borel's famous description of the cohomology ring of the full flag variety G/B [6]. For the reader's convenience, we briefly recall it: Borel showed that the cohomology ring of the full flag variety G/B is isomorphic to the ring of coinvariants of the Weyl group $W = N_G(T)/T$ of *G*

$$H^*(G/B,\mathbb{R}) \simeq \operatorname{Sym}(M_{\mathbb{R}})/\operatorname{Sym}(M_{\mathbb{R}})_{>0}^W,$$

where the action of *W* on the ring of polynomials on the Lie algebra of *T* is given by the coadjoint action on $M_{\mathbb{R}}$. In particular, $H^*(G/B, \mathbb{R})$ is generated in degree 2.

Let $U \subseteq B$ be the unipotent radical of B and let $U \subseteq H \subseteq B$ be any closed subgroup of G. Then, one has $N_G(H) = B$, and hence, there is a natural H/B-principal bundle

$$p: E = G/H \to G/B.$$

The group B/H is an algebraic torus (it is a factor group of $B/U \simeq (\mathbb{C}^*)^{\mathrm{rk}(G)}$), so $p: E \to G/B$ is a torus principal bundle. In what follows we identify B/U with the maximal torus T via natural projection $T \to B/U$.

As in Sect. 9, we can apply Theorem 10.4 to compute the cohomology ring $H^*(E_X, \mathbb{R})$ of the toric bundle associated with a smooth projective fan Σ in $M_{\mathbb{R}}(B/H)$. Denote by f_W the top homogeneous part of the Weyl polynomial on $M_{\mathbb{R}}(T)$. Then, Proposition 10.3 becomes

Proposition 10.5 Let G, T and f_W be as before. Then, for any $\lambda \in M_{\mathbb{R}}(T)$, we have

$$c(\lambda)^s = s! \cdot f_W(\lambda)$$
 where $s = \dim_{\mathbb{C}}(G/B)$.

Let *L* be the image of $M_{\mathbb{R}}(B/H)$ in $M_{\mathbb{R}}(T)$ under the pull-back of characters and let μ be a translation invariant Lebesgue measure on *L* normalized with respect to the lattice M(B/H).

Theorem 10.6 Consider the torus bundle $E = G/H \rightarrow G/B$, for $U \subseteq H \subseteq B$. Let $\Sigma \subseteq M_{\mathbb{R}}(B/H)$ be a smooth projective fan with associated toric variety X. Then, the cohomology ring $H^*(E_X, \mathbb{R})$ is given by

$$H^{*/2}(E_X,\mathbb{R})\simeq \operatorname{Diff}(\mathcal{AP}_{L,\Sigma})/\operatorname{Ann}(I),$$

Deringer

where $I(\Delta) = \int_{\Delta} f_W(\lambda) d\mu$. Furthermore, the ring of conditions $C^*_{\mathbb{R}}(G/U)$ is given by

 $C^*_{\mathbb{R}}(G/U) \simeq \operatorname{Diff}(\mathcal{AP}_L)/\operatorname{Ann}(I).$

Proof The theorem follows immediately from Corollary 9.4 and Proposition 10.5. \Box

10.3 Relation to String Polytopes

In the next two subsections, we continue to examine the example of toroidal horospherical varieties over full flag varieties. In this case, Theorems 10.4 and 10.6 have beautiful reformulations. For simplicity of the exposition, we will start with the case $G = SL_n$ and $H = U \subseteq SL_n$.

For λ a dominant weight, let $GZ_{\lambda} \subseteq \mathbb{R}^{N=n(n-1)/2}$ be the corresponding *Gelfand–Zetlin polytope* [15]. From the defining inequalities, it can be straightforwardly deduced that Gelfand–Zetlin polytopes are additive

 $GZ_{\lambda+\mu} = GZ_{\lambda} + GZ_{\mu}$ for any dominant weights λ, μ .

By linearity, we extend the definition of Gelfand–Zetlin polytopes to all of the positive Weyl chamber C^+ , i.e., we define GZ_{λ} for any $\lambda \in C^+$: if λ is in the positive Weyl chamber, we can write $\lambda = \sum_{i=1}^{r} a_i \cdot \mu_i$ for positive real numbers a_i and dominant (integral) weights μ_i ; set $GZ_{\lambda} = \sum_{i=1}^{r} a_i \cdot GZ_{\mu_i}$ (Minkowski addition). By the additivity of Gelfand–Zetlin polytopes, it follows that this definition is well defined.

The Gelfand–Zetlin polytopes form a linear family of polytopes, a concept which we recall from [30, Definition 1.2]. Let V, W be two real vector spaces (not necessarily finite-dimensional) and let $C \subseteq V$ be a full-dimensional convex cone. Note that the cone C is not necessarily closed. We only assume that $c_1x_1 + c_2x_2 \in C$ for any $x_1, x_2 \in C$ and all non-negative real numbers $c_1, c_2 \geq 0$. An \mathbb{R} -linear map $\Delta: C \to \mathcal{P}^+_W$ is called a *linear family of polytopes* where "*linearity*" is explained in the following folklore result:

Proposition 10.7 A map $\Delta: C \to \mathcal{P}_W^+$ is called linear if either of the following equivalent conditions is satisfied:

- (1) $\Delta(c_1x_1 + c_2x_2) = c_1\Delta(x_1) + c_2\Delta(x_2)$ for any $x_1, x_2 \in C$ and all non-negative real numbers $c_1, c_2 \geq 0$.
- (2) There is a (unique) linear map V → P_W, such that its restriction to C coincides with Δ(·).

For the reader's convenience, we include a proof for Proposition 10.7.

Proof The implication "(2) \Rightarrow (1)" is straightforward. It remains to show the reverse implication.

Choose a (Hamel) basis of V, say $\{v_i\}_{i \in I}$. Since C is full dimensional, we may (and will) assume that $\{v_i\}_{i \in I} \subseteq C$. We define a linear map $F: V \to \mathcal{P}_W$ by setting $F(v_i) = \Delta(v_i)$ for $i \in I$ and show that $F|_C = \Delta$. Let $x \in C$ expressed as x = $\sum_{i \in J} \lambda_i x_i$ for a finite subset $J \subseteq I$ and $\lambda_i \in \mathbb{R}$. Consider the partition of J

$$J_{-}:=\{j \in J : \lambda_{j} < 0\} \text{ and } J_{+}:=\{j \in J : \lambda_{j} > 0\}.$$

Then, $x + \sum_{j \in J_{-}} (-\lambda_j) v_j = \sum_{j \in J_{+}} \lambda_j v_j$, and thus

$$\begin{split} \Delta\left(x + \sum_{j \in J_{-}} (-\lambda_{j})v_{j}\right) &= \Delta\left(\sum_{j \in J_{+}} \lambda_{j}v_{j}\right) \Rightarrow \Delta(x) + \sum_{j \in J_{-}} (-\lambda_{j})\Delta(v_{j}) \\ &= \sum_{j \in J_{+}} \lambda_{j}\Delta(\lambda_{j}) \Rightarrow \Delta(x) = \sum_{j \in J} \lambda_{j}\Delta(v_{j}), \end{split}$$

where the second implication follows by our assumption that (1) holds. The statement follows by the observation that $F(x) = \sum_{j \in J} \lambda_j \Delta(v_j)$.

Example 10.8 It is straightforward to show that Gelfand–Zetlin polytopes GZ: $(C^+)^{\circ} \rightarrow \mathcal{P}^+_{\mathbb{R}^N}$ form a linear family of polytopes. Here, $(C^+)^{\circ}$ denotes the interior of the positive Weyl chamber.

For a linear family of polytopes $\Delta \colon C \to \mathcal{P}^+_W$, we construct its $lift \widetilde{\Delta} \colon \mathscr{C} \to \mathcal{P}^+_{V \oplus W}$, where \mathscr{C} denotes the following convex cone (C° denotes the interior of C):

$$\mathscr{C} := \{ Q \in \mathcal{P}_V^+ \colon Q \subseteq C^\circ \}.$$

Proposition 10.9 *C* is a full-dimensional convex cone.

Proof A straightforward calculation shows that \mathscr{C} is a convex cone. To show that $\mathscr{C} \subseteq \mathcal{P}_V$ is full dimensional, let $Q = Q_1 - Q_2 \in \mathcal{P}_V$, where $Q_1, Q_2 \in \mathcal{P}_V^+$. Since $C \subseteq V$ is full dimensional, there is a translation vector $v \in V$, such that $Q_1 + v, Q_2 + v \subseteq C$. The statement follows from the observation that $Q = (Q_1 + v) - (Q_2 + v)$.

We define the lift $\widetilde{\Delta} \colon \mathscr{C} \to \mathcal{P}^+_{V \oplus W}$ of the linear family of polytopes $\Delta \colon C \to \mathcal{P}^+_W$ by setting for any $Q \in \mathscr{C}$

$$\overline{\Delta}(Q) := \{ (x, y) \in V \oplus W : x \in Q, y \in \Delta(x) \}.$$

Proposition 10.10 Suppose V is finite-dimensional. Then, for any $Q \in \mathscr{C}$, $\widetilde{\Delta}(Q)$ is a polytope, i.e., $\widetilde{\Delta}(Q) \in \mathcal{P}^+_{V \oplus W}$. Furthermore, $\widetilde{\Delta} \colon \mathscr{C} \to \mathcal{P}^+_{V \oplus W}$ is linear in the sense of Proposition 10.7.

Proof Let $Q \in \mathscr{C}$. Consider the *H*-representation of *Q*, that is

$$Q = \bigcap_{i=1}^{s} \{x \in V : \langle x, n_i \rangle \le b_i\} \text{ where } n_i \in V^* \text{ are the facet normals and } b_i \in \mathbb{R}.$$

To determine the *H*-representation of $\widetilde{\Delta}(Q)$, we also need the *H*-representation of $\Delta(x)$ for $x \in C^{\circ}$. By [30, Proposition 1.3], the polytopes $\Delta(x)$ for $x \in C^{\circ}$ share the same normal fan (indeed, Kaveh and Villella assume *C* to be polyhedral, but their argument can be easily generalized for arbitrary convex cones). In particular

$$\Delta(x) = \bigcap_{j=1}^{\sigma} \{ y \in W \colon \langle y, v_j \rangle \le \beta_j(x) \}$$

where $v_j \in W^*$ are the facet normals and $\beta_j(x) \in \mathbb{R}$

Since for $x, x' \in C^{\circ}$, $\Delta(x)$, $\Delta(x')$, and $\Delta(x) + \Delta(x')$ share the same normal fan (see for instance [37, Proposition 7.12]), it straightforwardly follows from the linearity of Δ that the $\beta_j: C^{\circ} \to \mathbb{R}$ are linear in the sense of Proposition 10.7. In particular, β_j can be uniquely extended to a linear function on *V*, i.e., $\beta_j \in V^*$.

From the above, it follows that $\widetilde{\Delta}(Q) \subseteq V \oplus W$ is a polytope as $\widetilde{\Delta}(Q)$ is bounded with *H*-representation:

$$\widehat{\Delta}(Q) = \{(x, y) \in V \oplus W \colon \langle x, n_i \rangle \le b_i \text{ for } i = 1, \dots, s \text{ and } \langle y, \nu_j \rangle \le \langle x, \beta_i \rangle \text{ for } j = 1, \dots, \sigma \}.$$

Hence, $\widetilde{\Delta}(Q) \in \mathcal{P}_{V \oplus W}^+$. Using the linearity of Δ (in the sense of Proposition 10.7), a straightforward calculation shows that $\widetilde{\Delta} \colon \mathscr{C} \to \mathcal{P}_{V \oplus W}$ is linear too.

We denote the composition of the (linear) lift $\widetilde{\Delta} : \mathscr{C} \to \mathcal{P}^+_{V \oplus W}$ with the volume polynomial on $\mathcal{P}_{V \oplus W}$ by $\widetilde{\text{Vol}}$. The polynomial $\widetilde{\text{Vol}}$ has degree $\dim(V) + \dim(W)$.

Let us apply this construction of linear families of virtual polytopes to the linear family of Gelfand–Zetlin polytopes $GZ: C^+ \to \mathcal{P}^+_{\mathbb{R}^N}$. Let $B, T \subseteq SL_n$ and M = M(T) be as before. The polynomial $\widetilde{\text{Vol}}: \mathcal{P}_{M\mathbb{R}} \to \mathbb{R}$ has degree dim $(M_{\mathbb{R}}) + N = \dim_{\mathbb{C}}(E)$, and thus, its degree coincides with the degree of the polynomial $I(\Delta)$. Indeed, these two polynomials coincide up to a constant scalar multiple. We get the following result.

Theorem 10.11 Consider the torus bundle $E = SL_n / U \rightarrow SL_n / B$. Let $\Sigma \subseteq M_{\mathbb{R}}$ be a smooth projective fan and let $\Delta \in \mathcal{P}_{M_{\mathbb{R}},\Sigma}$. Then, we have

$$\rho(\Delta)^{n+N-1} = (n+N-1)! \cdot \widetilde{\text{Vol}}(\Delta),$$

where $N = \dim(\operatorname{SL}_n / B)$.

Proof Since Gelfand–Zetlin polytopes are linear in λ and $|GZ_{\lambda} \cap M| = \dim(V_{\lambda})$, we have $Vol(GZ_{\lambda}) = f_W(\lambda)$. Hence, by Proposition 10.5

$$c(\lambda)^N = N! \cdot \text{Vol}(\text{GZ}_{\lambda}),$$

and hence, $\int_{\Delta} c(\lambda)^N d\mu = N! \cdot \widetilde{\text{Vol}}(\Delta)$. The statement follows by Theorem 10.4. \Box

As an immediate corollary of Theorem 10.11, we obtain the following result.

Theorem 10.12 Consider the torus bundle $E = SL_n / U \rightarrow SL_n / B$. Let $\Sigma \subseteq M_{\mathbb{R}}$ be a smooth projective fan with corresponding toric variety X. Then, the cohomology ring $H^*(E_X, \mathbb{R})$ is given by

$$H^{*/2}(E_X, \mathbb{R}) \simeq \operatorname{Diff}(\mathcal{P}_{M_{\mathbb{R}}, \Sigma}) / \operatorname{Ann}(\operatorname{Vol}).$$

The ring of conditions $C^*_{\mathbb{R}}(\mathrm{SL}_n/U)$ is given by

$$C^*_{\mathbb{R}}(\mathrm{SL}_n/U) \simeq \mathrm{Diff}(\mathcal{P}_{M_{\mathbb{R}}})/\mathrm{Ann}(\mathrm{Vol}).$$

Proof The statements follow from Theorem 10.11 and Remark 9.5 as the map $c: M_{\mathbb{R}}(T) \to H^2(\mathrm{SL}_n/B, \mathbb{R})$ is surjective by Borel's theorem.

10.4 General Toric Bundles over a Full Flag Variety G/B

In general, suppose *G* is an (arbitrary) connected semisimple simply connected group and let $U \subseteq B \subseteq G$ be as above. In this section, we will give a general construction for toric bundles $G/H \rightarrow G/B$, with $U \subseteq H \subseteq B$. We use the same letter *N* to denote the (complex) dimension of the full flag variety G/B. As above, M = M(T)denotes the character lattice of *T*.

For any character λ in the positive Weyl chamber, one defines a *string polytope* $S_{\lambda} \subseteq \mathbb{R}^{N}$, such that

$$|S_{\lambda} \cap M(T)| = \dim(V_{\lambda}).$$

The construction of string polytopes depends on the choice of a reduced expression \underline{w}_0 of the longest element w_0 in the Weyl group, so fix some reduced expression \underline{w}_0 . There exists a convex cone $C_{\underline{w}_0} \subseteq M_{\mathbb{R}} \oplus \mathbb{R}^N$ called *weighted string cone*, such that

$$S_{\lambda} = \pi^{-1}(\lambda) \cap \mathcal{C}_{w_0},$$

where $\pi : M_{\mathbb{R}} \oplus \mathbb{R}^N \to M_{\mathbb{R}}$ is the natural projection (see [10, 31]). Like Gelfand– Zetlin polytopes, string polytopes are homogeneous in λ , i.e., $S_{k\lambda} = kS_{\lambda}$. However, string polytopes S_{λ} are not necessarily additive, i.e., $S_{\lambda+\mu} = S_{\lambda} + S_{\mu}$ might not be true. As in the case of SL_n, we get $c(\lambda)^N = N! \cdot \operatorname{Vol}(S_{\lambda})$ (see also [1, Remark 3.9 (iii)] or [21, Theorem 2.5]).

By [30, Proposition 1.4], it follows that the map which sends weights to the corresponding string polytopes is piecewise linear with respect to a fan \mathcal{F} which subdivides the positive Weyl chamber. That is the restriction of the map $S: \mathbb{C}^+ \to \mathcal{P}_{\mathbb{R}^N}; \lambda \mapsto S_{\lambda}$ to every cone of \mathcal{F} is linear. In particular, if σ is a full-dimensional cone of \mathcal{F} , then $S: \sigma \to \mathcal{P}_{\mathbb{R}^N}^+$ is a linear family of polytopes. By the construction from above, we obtain its corresponding lift $\mathcal{P}_{M_{\mathbb{R}}} \to \mathcal{P}_{M_{\mathbb{R}} \oplus \mathbb{R}^N}$ which gives rise to a polynomial Vol of degree dim $(M_{\mathbb{R}}) + N = \dim(G/U)$.

Now, let W be the image of $M_{\mathbb{R}}(B/H)$ in $M_{\mathbb{R}}(T)$ and let μ be the translation invariant measure on W normalized with respect to the lattice M(B/H). Since the

space of affine virtual polytopes \mathcal{AP}_W is a natural subspace of $\mathcal{P}_{M_{\mathbb{R}}(T)}$, their lifts are well-defined. In particular, for a virtual polytope $\Delta \in \mathcal{AP}_W$, its lift $\widetilde{\Delta}$ is a virtual polytope in a (dim W + N)-dimensional space. Let us denote by $\widetilde{\text{Vol}}$ its (dim W + N)-dimensional volume induced by μ . The same arguments as above yield generalizations of Theorems 10.11 and 10.12:

Theorem 10.13 Consider the torus bundle $E = G/H \rightarrow G/B$. Let $\Sigma \subseteq M_{\mathbb{R}}(B/H)$ be a smooth projective fan with associated toric variety X. Let $W \subseteq M_{\mathbb{R}}(T)$ be as before and let $\Delta \in \mathcal{AP}_{W,\Sigma}$ be an affine virtual polytope. Then, we have

$$\bar{\rho}(\Delta)^{r+N} = (r+N)! \cdot \widetilde{\text{Vol}}(\Delta),$$

where $r = \dim(M_{\mathbb{R}}(B/H)) = \dim(W)$. Moreover, the cohomology ring $H^*(E_X, \mathbb{R})$ is given by

$$H^{*/2}(E_X, \mathbb{R}) \simeq \operatorname{Diff}(\mathcal{AP}_{W,\Sigma})/\operatorname{Ann}(\widetilde{\operatorname{Vol}}),$$

and the ring of conditions $C^*(G/H)$ is given by

$$C^*_{\mathbb{R}}(G/H) \simeq \operatorname{Diff}(\mathcal{AP}_W)/\operatorname{Ann}(\operatorname{Vol}).$$

In this case, the ring of conditions $C^*_{\mathbb{R}}(G/H)$ coincides with the ring of complete intersections defined in Ref. [24] and our description of $C^*_{\mathbb{R}}(G/H)$ is analogous to the one given there.

Remark 10.14 Note that the lift of the linear family of string polytopes depends on the choice of a maximal cone σ of \mathcal{F} . However, since for λ in the positive Weyl chamber the volume of S_{λ} coincides (up to a constant multiple) with the degree of the corresponding line bundle \mathcal{L}_{λ} , the lifted volume polynomial Vol is independent of the choice of σ of \mathcal{F} . Indeed, for polytopes Q in the interior of the positive Weyl chamber, $\widetilde{\text{Vol}}(Q) = \int_{Q} \frac{1}{N!} c(\lambda)^N d\lambda$.

10.5 Projective Bundles

Here, we prove the classical description of the cohomology ring of projective bundles using our computations of cohomology rings of toric bundles (see for instance [14, Theorem 3.3]).

Theorem 10.15 Let *B* be a smooth orientable closed manifold and let $V \rightarrow B$ be a complex vector bundle of rank *n*. Consider the projective bundle compactification $E_V = \mathbb{P}(E \oplus \mathcal{O})$ of *V* (\mathcal{O} denotes the trivial line bundle on *B*). Then, the cohomology ring of $\mathbb{P}(E \oplus \mathcal{O})$ can be computed as

$$H^*(E_V, \mathbb{R}) \simeq \frac{H^*(B, \mathbb{R})[t]}{t^{n+1} + c_1(V \oplus \mathcal{O}) \cdot t^n + \dots + c_n(V \oplus \mathcal{O}) \cdot t + c_{n+1}(V \oplus \mathcal{O})}$$

Here, $c_1(V \oplus \mathcal{O}), \ldots, c_{n+1}(V \oplus \mathcal{O})$ denote the Chern classes of the vector bundle $V \oplus \mathcal{O}$ of rank n + 1 over B.

Proof By the Leray–Hirsch theorem, $H^*(E_V, \mathbb{R}) \simeq H^*(B, \mathbb{R}) \otimes H^*(\mathbb{P}^n, \mathbb{R})$, and in particular, there is a surjection

$$\phi: H^*(B, \mathbb{R})[t] \to H^*(E_V, \mathbb{R}).$$

We show that $\ker(\phi) = \langle P(t) = t^{n+1} + c_1(V \oplus \mathcal{O}) \cdot t^n + \dots + c_n(V \oplus \mathcal{O}) \cdot t + c_{n+1}(V \oplus \mathcal{O}) \rangle$. Since by Leray–Hirsch

$$\dim_{\mathbb{R}}\left(\frac{H^{*}(B,\mathbb{R})[t]}{t^{n+1}+c_{1}(V\oplus\mathcal{O})\cdot t^{n}+\cdots+c_{n}(V\oplus\mathcal{O})\cdot t+c_{n+1}(V\oplus\mathcal{O})}\right)$$
$$=\dim_{\mathbb{R}}(H^{*}(E_{V},\mathbb{R})),$$

it remains to show that $P(t) \in \ker(\phi)$. By the splitting principle, it is enough to show that $P(t) \in \ker(\phi)$ for vector bundles V that are direct sums of line bundles $V \simeq \mathcal{L}_1 \oplus \cdots \oplus \mathcal{L}_n$ on B.

In this case, let $E \to B$ be a toric bundle which corresponds to $V \simeq \mathcal{L}_1 \oplus \ldots \oplus \mathcal{L}_n$. Then, E_V is a toric bundle given by a fan Σ of the projective space \mathbb{P}^n . After an appropriate choice of coordinates in $N \simeq \mathbb{Z}^n$ (the lattice of 1-parameter subgroups), the fan Σ coincides with the normal fan of the standard simplex. More precisely, Σ has rays generated by $e_1, \ldots, e_n, e_{n+1}$, where

$$e_1 = (1, 0, \dots, 0), \dots, e_n = (0, \dots, 0, 1), e_{n+1} = (-1, \dots, -1),$$

and the cones of Σ are comprised by the conical hull of any subset of these rays.

By Theorem 3.8, the cohomology ring of E_V can be computed as

$$H^*(E_V, \mathbb{R}) \simeq \frac{H^*(B, \mathbb{R})[H_1, \dots, H_{n+1}]}{\langle H_1 \cdots H_{n+1} \rangle + \langle c(\lambda_i) - H_i + H_{n+1} : i = 1, \dots, n \rangle}$$
$$\simeq \frac{H^*(B, \mathbb{R})[H_{n+1}]}{(H_{n+1} + c(\lambda_1)) \cdots (H_{n+1} + c(\lambda_n))},$$

where $\lambda_1, \ldots, \lambda_n$ denotes the basis of M (the character lattice of the torus) dual to e_1, \ldots, e_n .

It is left to notice that since $V = \mathcal{L}_{\lambda_1} \oplus \ldots \oplus \mathcal{L}_{\lambda_n}$, the Chern classes of $V \oplus \mathcal{O}$ can be computed as

$$c_i(V \oplus \mathcal{O}) = s_i(c(\lambda_1), \ldots, c(\lambda_n)),$$

where s_i is the *i*th elementary symmetric polynomial in *n* variables. Therefore, the theorem follows from the fact that:

$$(H_{n+1} + c(\lambda_1)) \cdots (H_{n+1} + c(\lambda_n)) = H_{n+1}^{n+1} + s_1(c(\lambda_1), \dots, c(\lambda_n)) \cdot H_{n+1}^n + \dots + s_n(c(\lambda_1), \dots, c(\lambda_n)) \cdot H_{n+1}.$$

Acknowledgements The authors would like to thank Megumi Harada and Kiumars Kaveh for several helpful and inspiring conversations. The authors would also like to thank Michel Brion for his comments on the earlier version of this paper.

Funding The first author is supported by a Nottingham Research Fellowship from the University of Nottingham. The second author is partially supported by the Canadian Network for Research and Innovation in Machining Technology, Natural Sciences and Engineering Research Council of Canada Grant no. 156833-17. The third author is supported by EPSRC Early Career Fellowship EP/R023379/1. Open access funding provided by EPFL Lausanne.

Data availability Not applicable.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- 1. Alexeev, V., Brion, M.: Toric degenerations of spherical varieties. Sel. Math. New Ser. 10(4), 453 (2005)
- 2. Bialynicki-Birula, A.: Some theorems on actions of algebraic groups. Ann. Math. 98, 480–497 (1973)
- Bernstein, D.N.: The number of roots of a system of equations. Funkcional. Anal. i Priložen. 9(3), 1–4 (1975)
- Bernšteĭn, I.N., Gel'fand, I.M., Gel'fand, S.I.: Schubert cells, and the cohomology of the spaces G/P. Uspehi Mat. Nauk 28(3(171)), 3–26 (1973)
- Bernstein, D.N., Kušnirenko, A.G., Hovanskiĭ, A.G.: Newton polyhedra. Uspehi Mat. Nauk 31(3(189)), 201–202 (1976)
- Borel, A.: Sur la cohomologie des espaces fibrés principaux et des espaces homogènes de groupes de Lie compacts. Ann. Math. 2(57), 115–207 (1953)
- Brion, M.: Groupe de picard et nombres caractéristiques des variétés sphériques. Duke Math. J. 58(2), 397–424 (1989)
- Brion, M.: Piecewise polynomial functions, convex polytopes and enumerative geometry. In: Parameter Spaces (Warsaw, 1994), Volume 36 of Banach Center Publications, pp. 25–44. Polish Academic Science Institute of Mathematics, Warsaw (1996)
- Bott, R., Tu, L.W.: Differential Forms in Algebraic Topology. Graduate Texts in Mathematics, vol. 82. Springer, New York (1982)
- Berenstein, A., Zelevinsky, A.: Tensor product multiplicities, canonical bases and totally positive varieties. Invent. Math. 143(1), 77–128 (2001)
- Cox, D.A., Little, J.B., Schenck, H.K.: Toric Varieties. Graduate Studies in Mathematics, vol. 124. American Mathematical Society, Providence (2011)
- Davis, Michael W., Januszkiewicz, Tadeusz: Convex polytopes, coxeter orbifolds and torus actions. Duke Math. J. 62(2), 417–451 (1991)
- De Concini, C., Procesi, C.: Complete symmetric varieties. II. Intersection theory. In: Algebraic Groups and Related Topics (Kyoto/Nagoya, 1983), Volume 6 of Advanced Studies in Pure Mathematics, pp. 481–513. North-Holland, Amsterdam (1985)

- Fulton, W.: Intersection Theory, Volume 2 of Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics], 2nd edn. Springer, Berlin (1998)
- Gelfand, I.M., Cetlin, M.L.: Finite-dimensional representations of the group of unimodular matrices. Dokl. Akad. Nauk SSSR (N.S.) 71, 825–828 (1950)
- Griffiths, P., Harris, J.: Principles of Algebraic Geometry. Wiley Classics Library. Wiley, New York (1994). (reprint of the 1978 original)
- 17. Griffiths, P., Schmid, W.: Locally homogeneous complex manifolds. Acta Math. 123, 253-302 (1969)
- 18. Hatcher, A.: Algebraic Topology. Cambridge University Press, Cambridge (2002)
- 19. Hattori, Akio, Masuda, Mikiya: Theory of multi-fans. Osaka J. Math. 40(1), 1-68 (2003)
- Hofscheier, J.: The ring of conditions for horospherical homogeneous spaces. In: Proceedings of the Conference Interactions with Lattice Polytopes Magdeburg (2019)
- Kaveh, K.: Note on cohomology rings of spherical varieties and volume polynomial. J. Lie Theory 21(2), 263–283 (2011)
- Kazarnovskii, B., Ya.: Newton polyhedra and the Bezout formula for matrix-valued functions of finite-dimensional representations. Funct. Anal. Appl. 21(4), 319–321 (1987)
- Kaveh, K., Khovanskii, A. G.: Newton polytopes for horospherical spaces. Mosc. Math. J. 11(2), 265–283 (2011)
- Kaveh, K., Khovanskii, A. G.: Intersections of hypersurfaces and ring of conditions of a spherical homogeneous space. SIGMA Symmetry Integrability Geometry: Methods and Applications 16:Paper No. 016, 12 (2020)
- 25. Kleiman, S.L.: The transversality of a general translate. Compos. Math. 28, 287-297 (1974)
- Khovanskii, A., Limonchenko, I., Monin, L.: Cohomology rings of quasitoric bundles. arXiv preprint arXiv:2112.14970 (2021)
- Khovanskii, A., Monin, L.: Gorenstein algebras and toric bundles. arXiv preprint arXiv:2106.15562 (2021)
- Knop, F.: The Luna–Vust theory of spherical embeddings. In: Proceedings of the Hyderabad Conference on Algebraic Groups (Hyderabad, 1989), pp. 225–249. Manoj Prakashan, Madras (1991)
- 29. Kouchnirenko, A.G.: Polyèdres de Newton et nombres de Milnor. Invent. Math. 32(1), 1-31 (1976)
- Kaveh, K., Villella, E.: On a notion of anticanonical class for families of convex polytopes. arXiv preprint arXiv:1802.06674, (2018)
- 31. Littelmann, P.: Cones, crystals, and patterns. Transform. Groups 3(2), 145–179 (1998)
- Pukhlikov, A.V., Khovanskii, A.G.: Finitely additive measures of virtual polyhedra. Algebra i Analiz 4(2), 161–185 (1992)
- Pukhlikov, A.V., Khovanskiĭ, A.G.: The Riemann–Roch theorem for integrals and sums of quasipolynomials on virtual polytopes. Algebra i Analiz 4(4), 188–216 (1992)
- 34. Sankaran, P., Uma, V.: Cohomology of toric bundles. Comment. Math. Helv. 78(3), 540–554 (2003)
- Timashev, D.A.: Homogeneous Spaces and Equivariant Embeddings, Volume 138 of Encyclopaedia of Mathematical Sciences. Springer, Heidelberg (2011). (invariant theory and algebraic transformation groups, 8)
- Weyl, H.: Theorie der Darstellung kontinuierlicher halb-einfacher Gruppen durch lineare Transformationen III. Math. Z. 24(1), 377–395 (1926)
- Ziegler, G.M.: Lectures on Polytopes. Graduate Texts in Mathematics, vol. 152. Springer, New York (1995)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.